

TESE DE DOUTORADO

WELD BEAD GEOMETRY MODELING IN GMAW PROCESS BY MACHINE LEARNING TECHNIQUES AND DATA MINING PROCESS

Rogfel Thompson Martínez

Brasília, Julho de 2021

UNIVERSIDADE DE BRASÍLIA

FACULDADE DE TECNOLOGIA

UNIVERSIDADE DE BRASÍLIA Faculdade de Tecnologia

TESE DE DOUTORADO

WELD BEAD GEOMETRY MODELING IN GMAW PROCESS BY MACHINE LEARNING TECHNIQUES AND DATA MINING PROCESS

Rogfel Thompson Martínez

Prof. Sadek C. Absi Alfaro, Ph.D, FT/UnB Orientador

Prof. Guillermo Alvarez Bestard, Dr, FGA/UnB Co-orientador

FICHA CATALOGRÁFICA

PROCESS BY MACHINE LEARNING			
ederal] 2021.			
xvi, 97 p., 210 x 297 mm (ENM/FT/UnB, Doutor, Engenharia Mecânica, 2021).			
Tese de Doutorado - Universidade de Brasília, Faculdade de Tecnologia.			
AW Process			
forcement Learning			
lo (série)			

REFERÊNCIA BIBLIOGRÁFICA

THOMPSON, R. (2021). WELD BEAD GEOMETRY MODELING IN GMAW PROCESS BY MACHINE LEARNING TECHNIQUES AND DATA MINING PROCESS. Tese de Doutorado, Departamento de Engenharia Mecânica, Universidade de Brasília, Brasília, DF, 97 p.

CESSÃO DE DIREITOS

AUTOR: Rogfel Thompson Martínez TÍTULO: WELD BEAD GEOMETRY MODELING IN GMAW PROCESS BY MACHINE LEARNING TECHNIQUES AND DATA MINING PROCESS. GRAU: Doutor em Sistemas Mecatrônicos ANO: 2021

É concedida à Universidade de Brasília permissão para reproduzir cópias desta Tese de Doutorado e para emprestar ou vender tais cópias somente para propósitos acadêmicos e científicos. Os autores reservam outros direitos de publicação e nenhuma parte dessa Tese de Doutorado pode ser reproduzida sem autorização por escrito dos autores.

Rogfel Thompson Martínez Depto. de Engenharia Mecânica (ENM) - FT Universidade de Brasília (UnB) Campus Darcy Ribeiro CEP 70919-970 - Brasília - DF - Brasil

Agradecimientos

Quiero agradecer primeramente a mis padres, que me facilitaron el camino para llegar hasta aquí y ser lo que soy. Quiero agradecer a mi hermana, muy importante en mi vida, complemento muy inportante en mi camino. Quiero agradecer a todas las fuerzar de mis ancestros que hicieron que este camino pareciera más fácil, fuera bien vivido y enriquecedor. Quiero agradecer a mi Cuba, fuente de mis aprendizados de vida, fuerta matriz que no me permite detenerme; a todos mis profesores de vida.

Quiero agradecer a Armando Plasencia, a Zoila, a Mylen, a Eridania e sobre todo al Guille y al profesor Sadek, por creer en mi y confiar que era posible realizar esta investigación satisfactoriamente.

Quiero agradecer a Brasil y la forma amorosa que me abrió sus puertas. A todos los hermanos cubanos que conocí aquí. A todos los brasileros que hicieron que hicieron muy fácil adaptarme y conocer este maravilloso país, a: Denise, Alysson y Renata, a toda la galera de meditación en el parque; a todos los amigos del congreso de pedagogía 2018; a Filipe, Jaciara y toda su familia; al tio Raymundo; al de gitanos de las fogatas de luna llena; a las personas maravillosas que conocí en el "7 e Meio" y en el "Sou conveniente", en especial Rose por las colaboraciones; a todos los magníficos integrantes de mi grupo de capoeira angola "Nzinga", donde he crecido y continuo creciendo como humano. Quiero agradecer a las dindas de Iná.

Quiero agradecer a todas las personas que formaron o forman parte de mi vida. A Iná por lo mucho que me enseña.

Dedicatória

A Iná, mi nueva fuente de energía y vida ...

A mis abuelos, no presentes físicamente pero sé que me acompañan ...

A mis padres ...

ABSTRACT

Weld Bead Geometry Model in GMAW process by Machine Learning Techniques and Data Mining Process

The GMAW process has a non-linear behavior and has led many researchers to develop several studies on it. One of the main interests has been to optimize the process to develop better performance in industrial processes. Thus, current advances in image processing, predictive model, and intelligent modeling can help to optimize processes. These techniques can obtain good results in welding analysis. These techniques can be grouped into, techniques of machine learning, deep learning, and reinforcement learning, and data mining processes. They are responsible for the current advances in prediction, real-time image classification, and intelligent control. Its application in the welding area has the potential for a better study and analysis of processes, optimization of welding technologies, and better process controls. This research focuses on the objective of developing a weld bead geometry model in GMAW process by applying machine learning techniques and data mining process. As a result of the research, a deep learning model was obtained for the analysis of the arc, a predictive model of the process behavior, and a mode to optimize it. The methodologies developed with these models demonstrate a valid efficiency to be applied in real GMAW processes.

RESUMO

Modelamento da Geometria do Cordão da Solda no Processo GMAW Mediante Técnicas de Aprendizado de Máquina e Processo de Mineração de Dados

O processo GMAW possui um comportamento não linear, e tem levado muitos pesquisadores a desenvolverem diversos estudos sobre ele. Um dos principais interesses tem sido otimizar o processo para desenvolver melhor desempenho nos processos industriais. Dessa forma, os avanços atuais em processamento de imagem, modelo preditivo e modelagem inteligente podem ajudar a otimizar processos. Essas técnicas podem obter bons resultados na análise de soldagem. As técnicas se podem agrupar em: técnicas de aprendizado de máquina, aprendizado profundo, aprendizado por reforco e processos de mineração de dados. Eles são responsáveis pelos avanços atuais em predições, classificação de imagens em tempo real e controle inteligente. Sua aplicação na área de soldagem tem como potencial um melhor estudo e análise de processos, otimização de tecnologias de soldagem e melhores controles de processo. Esta pesquisa tem como objetivo desenvolver um modelo de geometria do cordão de solda no processo GMAW através da aplicação das técnicas de machine learning e do processo de mineração de dados. Como resultado da pesquisa, foi obtido um modelo de deep learning para a análise do arco, um modelo preditivo do comportamento do processo e um modo com o objetivo de otimizá-lo. As metodologias desenvolvidas com esses modelos demonstram uma eficiência válida para serem aplicadas em processos reais de GMAW.

SUMMARY

1	INTROD	UCTION	1
	1.1	GENERAL OBJECTIVE	2
	1.2	SPECIFICS OBJECTIVES	2
	1.3	SUMMARY OF CONTRIBUTIONS	2
	1.4	POTENTIAL APPLICATIONS	2
	1.5	PUBLICATIONS RESULTS	3
	1.6	DOCUMENT STRUCTURE	3
2	GMAW	PROCESS, DATA ANALYSIS AND MODELING CONCEPTUALIZATIONS	5
	2.1	GAS METAL ARC WELDING (GMAW)	5
	2.1.1	GMAW PARAMETERS	7
	2.1.2	GMAW PROCESS AS A COMPLEX SYSTEM	8
	2.2	DATA ANALYSIS	11
	2.3	Machine Learning Techniques	12
	2.3.1	DEEP LEARNING	14
	2.3.2	INTELLIGENT MODELING	16
	2.4	DATA MINING PROCESS	18
	0 F		10
	2.5	CHAPTER CONSIDERATIONS	13
3	2.5 Data A	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE	13
3	2.5 Data A States	NALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE OF-OF-ARTS AND METHODOLOGICAL PROPOSALS	20
3	2.5 DATA A STATES 3.1	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS	20
3	2.5 DATA A STATES 3.1 3.1.1	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS	20 21
3	2.5 DATA A STATES 3.1 3.1.1 3.2	CHAPTER CONSIDERATIONS NALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING	20 20 21 21
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS	20 20 21 21 22
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN)	20 20 21 21 22 23
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS?	20 20 21 21 22 23 24
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL	20 21 21 22 23 24 27
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS. SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE.	20 20 21 21 22 23 24 27 28
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE. METHODOLOGIC DIAGRAMS	20 21 21 22 23 24 27 28 29
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6 3.7	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE S-OF-ARTS AND METHODOLOGICAL PROPOSALS. SENSORS. VISION SENSOR. IMAGES PROCESSING . MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE. METHODOLOGIC DIAGRAMS. CHAPTER CONSIDERATIONS.	20 20 21 22 23 24 27 28 29 30
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6 3.7 EQUIPM	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE METHODOLOGIC DIAGRAMS CHAPTER CONSIDERATIONS MENT, MATERIALS AND DATA ACQUISITION OF THE EXPERIMENTS	20 20 21 21 22 23 24 27 28 29 30 30 32
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6 3.7 EQUIPM 4.0.1	CHAPTER CONSIDERATIONS ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS SENSORS VISION SENSOR IMAGES PROCESSING MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE METHODOLOGIC DIAGRAMS CHAPTER CONSIDERATIONS MATERIALS AND DATA ACQUISITION OF THE EXPERIMENTS MATERIALS	20 20 21 21 22 23 24 27 28 29 30 30 32 33
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6 3.7 EQUIPM 4.0.1 4.0.2	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS. SENSORS. VISION SENSOR. IMAGES PROCESSING . MODELING A WELDING PROCESS ARTIFICIAL NEURAL NETWORK MODELS (ANN) WHY IS IT NECESSARY TO TEST AND COMPARE DIFFERENT MODELS? INTELLIGENT CONTROL FUTURE PERSPECTIVE. METHODOLOGIC DIAGRAMS CHAPTER CONSIDERATIONS MENT, MATERIALS AND DATA ACQUISITION OF THE EXPERIMENTS MATERIALS VISUALIZATION TECHNIQUE.	20 20 21 21 22 23 24 27 28 29 30 30 32 33 33
3	2.5 DATA A STATES 3.1 3.1.1 3.2 3.3 3.3.1 3.3.2 3.4 3.5 3.6 3.7 EQUIPM 4.0.1 4.0.2 4.1	ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE G-OF-ARTS AND METHODOLOGICAL PROPOSALS	20 20 21 22 23 24 27 28 29 30 29 30 32 33 33 33 34

	4.1.2 4 2	EXPERIMENTS 3	41 42
_	4. <i>C</i>	CHAFTER CONSIDERATIONS	42
5	RESUL	TS OF THE TWO METHODOLOGIES PROPOSALS FOR WELD BEAD	
	GEOME		43
	5.1		43
	5.1.1	SHORT CIRCUIT AND DROP DETECTION	43
	5.1.2	DROP VOLUME METHODOLOGY: CALCULATE VOLUME	47
	5.1.3	FREQUENCY DATASET CHARACTERISTICS	50
	5.2	PREDICTIVE MODEL FOR GMAW PROCESS	50
	5.2.1	RESULT OF VOLUME DATASET	51
	5.2.2	RESULT OF FREQUENCY DATASET	52
	5.3	SIMULATION OF WELD BEAD GEOMETRY FOR GMAW PROCESS	52
	5.3.1	TRAIN WELD BEAD GEOMETRY MODELING	53
	5.3.2	DROP VOLUME METHODOLOGY: SIMULATION	56
	5.3.3	DROP FREQUENCY METHODOLOGY: SIMULATION	57
	5.4	CHAPTER CONSIDERATIONS	60
6	DESIGN	N AND DISCUSSIONS OF RESULTS	61
	6.1	DESIGN	61
	6.2	DISCUSSIONS	62
7	CONCL	USION	66
8	FUTUR	E WORKS	67
BI	BLIOGI	RAPHIC REFERENCES	68
AF	PENDIC	ES	84
A	WELD-	BEAD GEOMETRY CALC CODE	85
в	Convo	LUTIONAL NEURAL NETWORK CODE	89
С		TED WIRE CALC CODE	90
D	Predic		92
Е	REINFO	DRCEMENT LEARNING CODE	94

FIGURES LIST

2.1	GMAW Circuit	6
2.2	Welding torch	6
2.3	Welding bead	7
2.4	IWMT problems	9
2.5	Welding trend: New analysis techniques	10
2.6	Welding trend: Old and new analysis techniques	11
2.7	New data analysis techniques diagram	12
2.8	Deep learning vs Neural network architecture	14
2.9	The human vision system	15
2.10	Convolutional neural networks	15
2.11	Building block of residual learning	16
2.12	Reinforcement Learning diagram	17
2 1	Diagram of songars	20
3.1 2.2	Diagram of sensors.	20
3.2 2.2	Diblic metric employies. Anthemister metric metric	24
3.3 2.4	Bibliometric analysis: Authors interrelationship	28
3.4 2.5	Drop Volume Methodologic Diagram	29
3.5	Drop Frequency Methodologic Diagram	29
4.1	Welding table	33
4.2	The principle of Shadowgraphy Back-lighting applied to welding	34
4.3	Experiment 1: Example of welding arc images	35
4.4	Experiment 2: Example of welding arc images	36
4.5	Experiment 1: Weld bead geometry	37
4.6	Experiment 2: Weld bead geometry	37
4.7	Sequence diagram of weld bead geometry calc	38
4.8	Experiment 1: Width geometry vs time	39
4.9	Experiment 1: Height geometry vs time	39
4.10	Experiment 1: Depth geometry vs time	39
4.11	Experiment 2: Width geometry vs time	40
4.12	Experiment 2: Height geometry vs time	40
4.13	Experiment 2: Depth geometry vs time	40
4.14	Experiment 3: Example of welding arc images	41
4.15	Experiment 3: Weld bead geometry	42
5.1	Short circuit and drop detachment	44
5.2	No short circuit and no drop detachment	45
5.3	ResNet-model confusion matrix	46

5.4	Algorithm for unmelted wire length calc	48
5.5	Umnelted wire length calculation	48
5.6	Volume distribution	49
5.7	Algorithm of reinforcement learning train	54
5.8	Algorithm of reinforcement learning model	55
5.9	Drop Volume Methodology: modeling geometric parameters	56
5.10	Drop Volume Methodology: control parameters	57
5.11	Drop Frequency Methodology: modeling geometric parameters	58
5.12	Drop Frequency Methodology: control parameters	59
6.1	Design of the proposed system	62
6.2	PC and IA system communication	63

TABLES LIST

3.1	Table articles with quality objective	26
3.2	Table articles with prediction objective	26
3.3	Table articles with control objective	27
4.1	Experiment 1: input parameters by time	34
4.2	Experiment 2: input parameters by time	35
4.3	Experiment 3: input parameters by time	41
5.1	Images distribution by class and process	46
5.2	Volume dataset	50
5.3	Frequency dataset	50
5.4	Minimus error calculation of each models(Volume dataset)	52
5.5	Minimum error calculation of each models(Frequency dataset)	52
6.1	Comparison of the two methodologies	64

1 INTRODUCTION

Welding is considered one of the most important processes of joining metals used in industries. It is used in the fabrication of simple structures, as well as components of a high degree of responsibility in chemical, petroleum, and nuclear industries. One of the welding techniques is Gas Metal Arc Welding (GMAW). This is a welding process in which heat is generated by an electric arc incorporating a continuous-feed consumable electrode that is shielded by an externally supplied gas. In addition, this welding technique can be applied in ferrous and non-ferrous materials. This is due to its versatility, relatively high productivity, reliability, and automation. GMAW is a stochastic process, and the interrelation between its inputs and outputs parameters have a non-linear behavior. Then, this is one of the main problems that researchers find when modeling a GMAW process. The balance between the initial parameters and metal transference modes defines the shape of the weld bead geometry. However the difficulty to control the relationship between these parameters has led to many studies that focus on making an analysis based on only one metal transfer mode. Another problem is that many investigations do not get a real-time analysis. Due to the slowness of the classic image processing techniques.

In practical welding production, welding conditions are often changing, such as the errors of pre-machining and fitting work-piece would result in differences of gap size and position, the change of work-piece heat conduction and dispersion during welding process would bring on weld distortion and penetration odds. The arc welding process contains complicated, stochastic and uncertain information. Monitoring the state of arc welding process is very important for predictive and controlling the welding process. Many sensing methods for welding process have been used in consideration of the disturbance from arc, high temperature, vibration, electromagnetic fields and the features of the process. Thus, in order to obtain the effectual features of arc welding process for real-time control of weld quality, various signal processing methods have been applied for information of welding process, such as processing algorithms for arc voltage, current, visual, optical, mechanical information.

It is apparent that machine learning techniques help to analyze and find solutions to many modeling problems. Machine learning uses the theory of statistics in building mathematical models, making inferences from larges samples of data. Deep learning techniques obtain excellent results in the classification of complex images. Further, other machine learning techniques, such as reinforcement learning are being used in intelligent control processes. These techniques have excellent performance in deployment environment after a hard model training. They, together, can accomplish a weld bead geometry model of GMAW process in real-time.

Furthermore, small computers with graphic cards can execute the weld bead geometry model in real-time. In addition to allowing the process to be faster than in a conventional computer, it will also make the equipment cheaper and lower energy consumption.

1.1 GENERAL OBJECTIVE

The aim of this work is to develop a weld bead geometry model of GMAW process by applying techniques of machine learning, deep learning, and reinforcement learning.

1.2 SPECIFICS OBJECTIVES

- To develop a real-time detection model of short circuit and droplet detachment in GMAW process using deep learning techniques.
- To develop a predictive model of the weld bead geometry of a GMAW process using machine learning techniques.
- To develop the weld bead geometry model of a GMAW process in real-time by applying reinforcement learning techniques.

1.3 SUMMARY OF CONTRIBUTIONS

This research contribute:

- In the proposal of two methodologies for weld bead geometry modeling of the GMAW process.
- In a new GMAW arc images analysis with deep learning technique in real-time.
- In the integration of transfer modes in one computational model for GMAW process.
- In a comparison of predictive GMAW model.
- In a new intelligent modeling process with reinforcement learning for GMAW process.

1.4 POTENTIAL APPLICATIONS

The research developed shows a series of models that both individually and together can be used for other studies of GMAW process. Also, the image processing techniques used can be useful for monitoring short circuits and drop detachment in a GMAW process. The predictive model represents a low-cost technique to identify the behavior of weld bead geometry of GMAW process. That way, this quality allows its use in teaching processes and simulation tests in the production industry. Indeed, the real-time modeling of the weld bead geometry to having the same potentialities of a predictive model is a precedent to define an intelligent control of GMAW process.

1.5 PUBLICATIONS RESULTS

Papers presented in conferences and published in conference proceedings.

- Thompson, R.; Absi Alfaro, S. C. A. and Martin, A. "Dynamic drop volume function and deep learning to obtain drop and molten volume in GMAW process", in *IV Conferencia Internacional de Soldadura y Unión de Materiales - ICONWELD 2018*. Lima, Perú. ISBN: 978-612-47965-0-0, pp. 136.
- Thompson, R.; Martin, A and Absi Alfaro, S. C. "Obtaining of drop and molten volume in GMAW process with a dynamic drop volume function and Deep Learning technique", in *XLIV CONSOLDA Congresso Nacional de Soldagem, 2018.* Uberlândia, Brazil.
- Thompson, R. and Absi Alfaro, S. C. "Intelligent control proposition on GMAW process with machine learning techniques", in *COB2019 Conference*, 2019. DOI: 10.26678/ABCM.COBEM2019.COB2019-0263.
- Thompson Martínez, R.; Alvarez Bestard, G.; Martins Almeida Silva, A.; and Absi Alfaro, S. C.; "Analysis of GMAW process with deep learning and machine learning techniques," J. Manuf. Process., vol. 62, pp. 695–703, Feb. 2021. https://doi.org/10.1016/j.jmapro.2020.12.052.
- Thompson Martinez, R., Alvarez Bestard, G., Absi Alfaro, S. C. (2021). Two Gas Metal Arc Welding process dataset of arc parameters and input parameters. Data in Brief, 35, 106790. https://doi.org/10.1016/j.dib.2021.106790.

Chapter of book.

 Thompson R. and Absi, S. C. (March 10th, 2020). "Data Analysis and Modeling Techniques of Welding Processes: The State-of-the-Art", Welding - Modern Topics, ISBN: 978-1-83881-896-8. DOI: 10.5772/intechopen.91184.

1.6 DOCUMENT STRUCTURE

This work is organized in six chapters, the bibliography references, and annexes.

Chapter 1 explained the research problem, as well as the objectives and their possible contribution.

Chapter 2 presents a conceptual analysis of the GMAW process, data analysis techniques, models process, and the interdisciplinary that they may possess.

Chapter 3 presents a bibliographic review of data analysis techniques in welding.

Chapter 4 presents the equipment and materials used, as well as the experimental design.

Chapter 5 presents the methodology developed in research based on the stages of data mining

processes. In each stage, the obtained results are explained.

Chapter 6 develops a proposal for the design of components necessary to apply the models obtained in an experimental lab.

Conclusions and future works present the general results of research and potential researches that can be developed from this one.

2 GMAW PROCESS, DATA ANALYSIS AND MODELING CONCEPTUALIZATIONS

This chapter describes the GMAW process, its characteristics, and fundamental parameters that will be analyzed in this research. Also, it presents a conceptual analysis and potentialities of an interdisciplinary area between welding processes, specifically the GMAW process, machine learning, and data mining process techniques. Moreover, the chapter explains the possible contributions of the computer science area in the welding process.

2.1 GAS METAL ARC WELDING (GMAW)

Welding is considered one of the most important processes of joining metals used in industries as shown in (Villani, Modenesi and Bracarense 2016). Because, it is used in the fabrication of simple structures, as well as components of a high degree of responsibility in chemical, petroleum, and nuclear industries. In other words, one of the welding techniques that contribute to the mentioned utilities is the Gas Metal Arc Welding (GMAW). It is a consumable electrode welding process that produces an electric arc between a weld pool and supplied electrode wire. The electric arc is its energy source to join metal pieces. The process can be performed either as an automated, as a manual hand-held process, or semi-automatic (Thomsen 2005). Furthermore, It can be applied to ferrous and non-ferrous materials. It is mainly due to its versatility, relatively high productivity, reliability, and ease of use and automation (ASM 1994, Scotti, A; Ponomarev, V 2008, Cary and Helzer 2005). This welding process is considered a highly non-linear, multi-parameters and time-varying system (Yan 2011). The basics equipment used for a typical GMAW semiautomatic setup is shown in Figure 2.1.

- (1) Welding torch: It contains wire electrode and shielding gas supplied (Figure 2.2). The electrode extension is the amount the end of the electrode wire sticks out beyond the end of the contact tube.
- (2) Workpieces: they are the metals to welding.
- (3) Welding Source: It is a constant voltage power source whose one terminal is connected with the welding torch and the other is connected to the workpiece through a clamping device.
- (4) Wire feed unit: it is the control to wire supply.
- (5) Electrode source: It is the metal wire which is used as the metal electrode in the GMAW welding.
- (6) Shielding gas supply: it provides a supply of shielding gas to the arc area.

Figure 2.1: GMAW Circuit



Source: (The Welding Master 2017)

Figure 2.2: Welding torch



Source: (EuroTech 2019)

2.1.1 GMAW parameters

GMAW parameters are the parameters involved in the welding process, whose changes have influences on the characteristics of heat, metal transfer and welding bead geometry (Cayo 2013). GMAW process offers some difficulty correcting welding parameters, this is mainly due to the relatively high number of parameters and above all, a strong interrelation between them (Scotti, A; Ponomarev, V 2008, Bingul and Cook 2006). In this research, some parameters are selected for GMAW process analysis. It is valuable to note the parameters are those that can be adjusted during the process, they are wire feed speed, welding velocity, and voltage. The adjustable parameters are those that make it possible to control some of them in the process. They are also known as input parameters. Add to this the arc parameters are constituted by those phenomena produced by the electric arc. Some experiments are analyzed as output parameters. These parameters were selected because they are those that can be estimated through arc-image processing. Some arc parameters are unmelted wire length, drop volume, melted wire volume, short circuit frequency, and drop frequency. Weld-bead parameters are weld-bead depth, weld-bead width, and weld-bead height. The bead parameters consist of geometric characteristics as shown in the Figure 2.3.

Figure 2.3: Welding bead



Source: (Pinto-Lopera et al. 2016)

GMAW process, being a process that uses consumable wire, is mainly characterized by metal transfer modes. Thus the metal transfer modes are influenced by GMAW input parameters, characteristics of materials, and components used. Metal transfer in the GMAW process can be grouped into three modes:

(1) Short-circuiting metal transfer is a mode of metal transfer whereby a continuously fed solid or metal-cored wire electrode is deposited during a repeated electrical short-circuiting. The short-circuiting metal transfer mode is the low heat input mode of metal transfer for GMAW. Then, the metal transfer occurs when the electrode is electrically shorted with the base material or molten puddle. Central to the successful operation of short-circuiting transfer is the diameter of electrode, the shielding gas type, and the welding procedure employed (The Lincoln Eletric Company 2014).

- (2) **Globular metal transfer** is a GMAW mode of metal transfer whereby a continuously fed solid or metal-cored wire electrode is deposited in a combination of short-circuits and gravity-assisted large droplets. The larger droplets are irregularly shaped. During the use of all metal-cored or solid wire electrodes for GMAW, there is a transition between short-circuiting transfer ends and globular transfer begins. Globular transfer characteristically gives the appearance of large irregularly shaped molten droplets that are larger than the diameter of the electrode. The process at this current level is difficult to control and spatter is severe. Gravity is instrumental in the transfer of the large molten droplets with occasional short-circuits (The Lincoln Eletric Company 2014).
- (3) **Spray metal transfer** is the higher energy mode of metal transfer whereby a continuously fed solid or metal-cored wire electrode is deposited at a higher energy level, resulting in a stream of small molten droplets. The droplets are propelled axially across the arc. For most of the diameters of filler metal alloys, the change to spray transfer takes place at the globular to spray transition current (The Lincoln Eletric Company 2014).

Indeed the variability of these modes and the inaccuracy of the values under which they occur are important elements that define the complexity of GMAW process. This makes the analysis of the arc a very important element that determines what can happen at the end of the process.

2.1.2 GMAW process as a complex system

Despite the great advantages of the GMAW process. It requires a careful setting of process parameters to avoid fusion defects, especially on thicker base metals. Careful configurations are ideal for a wide range of industrial welding requirements. Most common welding problems fall into the category of improper weld bead profile, As was shown in diversifying investigations such as (Moncayo Torres 2013, Giron Cruz 2014, Alvarez Bestard and Absi Alfaro 2018) exposed. In certain industrial applications, it is required to obtain specific dimensions from the weld bead geometry. Meeting these objectives requires an excellent understanding and enough experience, which few specialists have, as (Scotti, A; Ponomarev, V 2008) said.

Welding defect is defined as any flaw compromises the usefulness of any products, the irregularities in the weld metal produced due to incorrect welding parameters or wrong welding procedures or wrong combination of metal and non-metal plates. In the welding process there are many wear or defects occur. These wear factor may include oversize work piece, casting blow holes in the work piece, thermal and mechanical properties, and variation of different metal work piece hardness. The weld defects occur during welding due to the complexities of the welding processes, moisture in the air and the unpredictable factors (Mathers 2002, Rajeev Kumar and Dr. Vandana Somkuva 2015).

The welding process in general can be viewed as a complex system that has multiple inputs, multiple outputs, and multiple disturbances. This kind of system adds difficulty in determining the correct set of input values to achieve the desired outputs. The history of inputs to outputs can help to optimize the adjustment of input values to improve the speed taken to converge at the desired

levels of outputs. A non-linear model may be obtained to correlate outputs and inputs values as was showed by (Zhang YuMing 2008). Hence, (Chen and Lv 2014) suggested a modern welding manufacturing technology. The intelligentized welding manufacturing technology (IWMT) is mainly related to key intelligent technical elements. The studies establish the foundation work of intending researches and applications on intelligentized technologies for modern welding manufacturing. IWNT promotes systematization research for forming an effective combination of modern welding manufacturing, computer science, and artificial intelligence technology. In addition, the development of modern welding technology is changing from traditional handicraft to modern science manufacturing. They also show some key scientific and technical problems in IWMT. Figure 2.4 shows three of them that will be analyzed in this research.

Figure 2.4: The key scientific and technical problems in IWMT.



Source: (Chen and Lv 2014)

As of present, computer science has the potentials to solve these three problems. Computer science field had great results with the new technique applications of data analysis, learning models, and intelligent control. Data analysis objectives indicate non-trivial features on amount large data. Due to the increase and complexity data has been developed for more efficient data analysis techniques. Welding process can be analyzed from this point of view. As a result, welding process analysis with new techniques is nothing more than continuity in the development of welding analysis processes. This interdisciplinarity is one of the necessary contributions proclaimed by called industry 4.0, like (Haffner et al. 2017, Jiang, Zhang and Wang 2017, Chong, Ramakrishna and Singh 2018) shown. The industrial 4.0 refers to the next manufacturing generation, where automation technology will be improved by self-optimization and intelligent feedback (Tuominen 2016). For this reason, the application of the most recent data analysis

techniques and processes can contribute to better modeling and monitoring of welding processes. This new data analysis technique can be groups in Data Mining process and machine Learning techniques.

An inquiry conducted in the Web of Science from 2011 to October, 3rd 2018 shows that there is a growing trend of these new data analysis techniques in welding process researches, Figure 2.5 shows this trend. Whereas when comparing with the investigations on the models welding process, the growth is still very small, as appear in Figure 2.6. This confirmed a need to encourage research among these areas and greater socialization of results.

Figure 2.5: Cited per year on welding (Web of Science (Analytics 2018))



Data Analysis Trend (cited per year)

Source: Produced by author



2.2 DATA ANALYSIS

From Shannon's contributions to information theory (Weaver 1949), data analysis had increasing importance for technological, social, and scientific processes. Information theory is a branch of applied mathematics that involves the quantification of information as expressed (Liu et al. 2012). Entropy and information-theory principles are still widely used in data analysis in several areas as shows (Qi and Guo 2014, Varadan and Anastassiou 2006, Wollstadt et al. 2014, Dishion et al. 2004, Rhea et al. 2011, Seyyed and Mohammad 2011). An important objective of data analysis is to reveal and to indicate diverse, non-trivial features in data. For this reason, welding process can be analyzed from this point of view.

The contemporary techniques of data analysis can join in machine learning techniques as shown (Hernandez Orallo, Ramirez Quintana and Ferri Ramirez 2004, Yu and Deng 2011, Marsland 2015, Bell 2015, Casalino 2018), in data mining process as shown (Hirji 1999, Norton 1999, Olson and Delen 2008, Piatetsky 2014, Chambers, Doig and Stokes-Rees 2017) and in intelligent control process by machine learning and reinforcement learning techniques exemplified by (Chi et al. 2019, Huang et al. 2019, Woods and La 2019). The interrelation of these areas and their origins are presented in Figure 2.7.



Figure 2.7: Origin diagram of the new data analysis techniques

2.3 MACHINE LEARNING TECHNIQUES

In 1959, Arthur Samuel defined machine learning as, field of study that gives computers the ability to learn without being explicitly programmed (Bell 2015). Machine learning is one of the fast-growing areas of computer science, with far-reaching applications for data analysis. The term machine learning refers to the automated detection of meaningful patterns in data (Shalev-Shwartz and Ben-David 2014). Machine learning helps to find solutions to many problems in vision, speech recognition, and robotics. Thus, this uses the statistics theory in building mathematical models making inferences from a data sample. The role of computer science is twofold by (Alpaydin 2010). First, in training, we need efficient algorithms to solve the optimization problems, to store and process the massive amount of data we generally have. Second, once a model is learned, its representation and algorithmic solution for inference need to be efficient as well. The model may be predictive to make predictions in the future, or descriptive to gain knowledge from data or both. As an interdisciplinary field, machine learning shares common threads with the mathematical area of statistics, information theory, game theory, and optimization. It is naturally a subfield of computer science. Its goal is to program machines to

learn. In this sense, machine learning can be view as a branch of artificial intelligence (AI) since the ability to turn the experience into expertise or to detect meaningful patterns in complex sensory data is a cornerstone of human (and animal) intelligence (Shalev-Shwartz and Ben-David 2014). Two classification of machine learning algorithms are supervised learning and reinforcement learning. Supervised learning use a training set of examples with the target responses is provided. Based on this training set, the algorithm develops a model to respond correctly to all possible inputs. Reinforcement learning is used to solve interacting problems where the info observed up to time t is taken into account to decide which action to require at time t + 1. The algorithms that are categorized in this group can be used in optimization and intelligent control tasks (Marsland 2015). Some famous and best-used supervised machine learning techniques are:

- Linear discrimination: is a common statistical tool for modeling the relationship between some explanatory variables and some real-valued outcome. It is a discriminant based approach that estimates the parameters of the linear discriminant directly from a given labeled sample. A example of this technique is Linear Regression (LR) (Alpaydin 2010, Shalev-Shwartz and Ben-David 2014).
- Support vector machine (SVM): is one of the most popular algorithms in modern machine learning. It was introduced by Vapnik in 1992. It provides a very impressive classification performance on reasonably sized data sets. It consists of a vector representation of records, with a real component for each attribute (Marsland 2015).
- Nearest neighbor algorithms: are among the simplest of all machine learning algorithms. The idea is to memorize the training set and then to predict the label of any new instance on the basis of the labels of its closest neighbors in the training set (Shalev-Shwartz and Ben-David 2014).
- Bayesian networks: based on a set of variables or parameters, it is possible to predict outcomes based on probabilities. These variables are connected to each other that the resulting value of one variable will influence the output probability of another, hence the use of networked nodes. A Bayesian Network manages to combine probability theory with graph theory and provides a handy method for dealing with complexity and uncertainty (Bell 2015).
- Artificial neural network: is a computation model inspired by the structure of neural networks in the brain. In simplified models of the brain, it consists of a large number of basic computing devices (neurons) that are connected to each other in a complex communication network, through which the brain is able to carry out highly complex computations. Artificial neural networks are formal computation constructs that are modeled after this computation paradigm (Shalev-Shwartz and Ben-David 2014). Artificial neural networks (ANN) are the most popular artificial learning tool in computer science and other research disciplines (Casalino 2018).

These techniques have great applicability in the development of predictive models. They can be used in the development of a predictive model of the weld bead geometry of GMAW process. Consequently, for a good analysis of welding arc images, other techniques with better results in the image processing area would be necessary.

2.3.1 Deep Learning

Machine learning technicians are instrumental in signal processing investigations, although in 2006 a new area has emerged in automatic learning called *deep learning* (Yu and Deng 2011). Deep learning allows computer models to be composed of multiple-layers processing to represent the learning of these data with several layers of abstraction like it is shown in Figure 2.8. These methods are essential part of the research on speech recognition in states-of-the-arts (Mesnil et al. 2013), image recognition (Zhu et al. 2014, Pachitariu et al. 2013), object detection (Pachauri et al. 2014) and other domains as the human genome (LeCun, Bengio and Hinton 2015).

Figure 2.8: Deep learning vs Neural network architecture



2.3.1.1 Deep Learning in vision system

In past decades, traditional image-processing techniques were considered computer vision systems, but that is not accurate. A machine processing an image is completely different from that machine understanding what's happening within the image, which is not a trivial task. At the highest level, vision systems are pretty much the same for humans, animals, insects, and most living organisms. In the same way, they consist of a sensor or an eye to capture the image and a brain to process and interpret the image (Elgendy 2020), like is shown in Figure 2.9. Scientists were inspired by the human visual system and in recent years have done amazing research on visual ability with machines. Its works were initialized by Yann LeCun's paper in 1998 (Lecun et al. 1998). This paper reviews various methods applied to handwritten character

recognition and compares them on a standard handwritten digit recognition task.

Figure 2.9: The human vision system uses the eye and brain to sense and interpret an image

Human vision system



This new architecture is shown to outperform all other techniques. Convolutional neural networks (CNN), as is called, are specifically designed to deal with the variability of 2D shapes, like show the Figure 2.10. The convolutional and pooling layers have the objective of extracting the patterns that identify a group of images concerning others. The objective of the full connected layers is to transform these patterns to the corresponding classification, like it was expressed in (Krizhevsky, Sutskever and Hinton 2017).





Source: Produced by author.

2.3.1.2 Residual Neural Network (ResNet)

Deeper neural networks are more difficult to train. Residual Neural Network (ResNet) is to ease the training of networks that are substantially deeper than those used previously (He et al. 2015). ResNet is based on deep residual learning. It explicitly reformulates the layers as learning residual functions concerning the layer inputs, instead of learning unreferenced functions (He et al. 2015). The idea behind the above block is, instead of hoping every few stacked layers

directly fit a desired underlying mapping say H(x), It explicitly lets these layers fit a residual mapping F(x) = H(x) - x. Thus original mapping H(x) becomes F(x) + x, like it shows in Figure 2.11.

Figure 2.11: Building block of residual learning



Source: (He et al. 2015).

The advantage of adding this type of skip connection is because if any layer hurt the performance of architecture then it will be skipped by regularization. So, this results in training very deep neural network without the problems caused by vanishing/exploding gradient. In fact, the ResNet models were extremely successful. It won 1st place in the ImageNet Large Scale Visual Recognition Challenge (ILSVRC) 2015 classification competition with a error rate of 3.57% (Sik-Ho Tsang 2018). These advantages could be exploited in a model to detect short circuit and droplet detachment, which can help in obtaining arc parameters, like droplet frequency, short-circuit frequency, and an estimate of the molten-wire volume of GMAW process. Furthermore, these parameters can be input parameter for the predictive model of weld bead geometry.

2.3.2 Intelligent Modeling

The definition of intelligent modeling is very associated with reinforcement learning. Intelligent Modeling is a heuristic, autonomous, non-linear, and adaptive (with learning) controller. Intelligent Modeling is born with the intention of applying a control technique in information theory, computer science, artificial intelligence, in order to obtain good results in complex systems modeling and consolidated as a discipline (Santos 2011). Likewise, an application advantage of this control strategy, the mathematical model can be obtained with the techniques mentioned in the previous sections. As an optimization-based method, the q-learning algorithm can be applied. It can be used to solve optimal control problems (Li et al. 2018). Q-learning has shown a good control performance when exposed to time-varying external disturbances (Yin, Yu and Zhou 2018). Q-learning is an effective scheme for unknown dynamical systems because it does not require any knowledge of the system dynamics to solve optimal

control problems (Chun, Park and Choi 2018, Li et al. 2018). The q-learning algorithms are important pieces for reinforcement learning (RL) techniques. RL is a machine learning paradigms concerned with how software agents ought to take actions in an environment, so as to maximize some notion of cumulative reward (Sutton and Barto 2017). The decision-maker is an agent that interacts with the environment it's placed in, as shown in Figure 2.12. This process of selecting an action from a given state, transitioning to a new state, and receiving a reward happens sequentially over and over again, which creates something called a trajectory that shows the sequence of states, actions, and rewards (Deeplizard 2019).





Source: (Sutton and Barto 2017)

- State: The states are the values that must be controlled. These are the processes results that occur in the environment.
- Action: The actions are the agent executions to reach the control objective. the actions are the control variables.
- **Reward**: The reward is the prize value to reach the control objective.

The q-learning algorithm goal is to learn a policy, which tells an agent what action to take and under which circumstances(present state). Then, this algorithm can be used to model the weld bead geometry, using like environment the model predictive of weld bead geometry.

With only a good machine learning algorithm, you do not get a good learning model. Some analytic functions are often automated, but human setup prior to implementing procedures is required. Proper selection of data to include in searches is critical. Similarly, data transformation is often required. Too many variables produce too much output, while too few can overlook key relationships in the data (Olson and Delen 2008). Therefore, to obtain a good learning model, it is necessary to use a good data analysis process. Data mining is one of them. Data mining is defined as the process of discovering patterns in data (Witten and Frank 2005). Machine learning provides algorithmic techniques for data mining process.

2.4 DATA MINING PROCESS

Data mining is an interdisciplinary field that brings together techniques from machine learning, pattern recognition, statistics, databases, and visualization to address the issue of information extraction from large dataset (Hirji 1999). Data mining is a field of the intersection of computer science and statistics, used to discover patterns, prediction, and classification in the information bank. The main aim of the data mining process is to extract useful information from the dossier of data and mold it into an understandable structure for future use (Agarwal 2013). It has proven to be extremely effective in improving research in many areas. These articles (Pan and Yang 2010, Felzenszwalb et al. 2010, Shaw, Sicree and Zimmet 2010, Tamura et al. 2011), are the most cited articles in the area, with more than 2000 time citations from Web of Science. This demonstrates that the potentialities of data mining techniques can be applied in any area since its raw materials are given. Data mining requires a problem identification, along with a collection of data that can lead to better understanding, and computer modeling to provide statistical or other means of analysis (Olson and Delen 2008). All these requirements are defined in many similar processes and methodologies like:

- Knowledge Discovery in Databases: refers to the broad process of finding knowledge in data, and emphasizes the "high-level" application of particular data mining methods. It is of interest to researchers in machine learning, pattern recognition, databases, statistics, artificial intelligence, knowledge acquisition for expert systems, and data visualization (Piateski and Frawley 1991, Fayyad et al. 1996, Norton 1999).
- SEMMA Methodology: The SEMMA process was developed by the SAS Institute. The acronym SEMMA stands for Sample, Explore, Modify, Model, Assess, and refers to the process of conducting a data mining project (Olson and Delen 2008).
- CRISP-DM Methodology: which stands for Cross-Industry Standard Process for Data Mining, is an industry-proven way to guide your data mining efforts. As a methodology, it includes descriptions of the typical phases of a project, the tasks involved with each phase, and an explanation of the relationships between these tasks (Larose 2014, Piatetsky 2014).
- Data Science Process: is an agile, iterative data science methodology to deliver predictive analytics solutions and intelligent applications efficiently (Stanton 2012, Chambers, Doig and Stokes-Rees 2017).

Essentially, these processes and methodologies define some stages as shown in (Marbán, Mariscal and Segovia 2009). These stages can resume to:

- Research understanding, it shows the research objective.
- Data acquisition. It starts with an initial data collection and proceeds with activities to get familiar with the data.

- Data preparation. It covers all the activities required to construct the final dataset from the initial raw data for the model to develop.
- Modeling. Various modeling techniques are selected and applied. Their parameters are calibrated to optimal values.
- Evaluation. It evaluates the different models obtaining to select the best from this objective.
- Deployment. The objective is to deploy and use the Discovered Knowledge.

Each of the stages contains a series of good practices that help us to obtain a better final result (Chapman et al. 2000, Chambers, Doig and Stokes-Rees 2017). Interesting patterns come out from data mining practice. That is, besides some common properties, different perspectives of data mining put strong emphases on different aspects, like efficiency, effectiveness, and validity of process (Zhou 2003). Data mining process in GMAW process analysis should help to validate efficiency or effectiveness on data acquisitions, data preparations, and modeling. The methodology proposed in this research must be closely related to the steps of the referenced methodologies and processes. This will guarantee the tools for a better evaluation of each of the defined methodological stages.

2.5 CHAPTER CONSIDERATIONS

The transversality of data analysis techniques allows its use in various areas. The use of these analyses in welding processes is still new. Although, there are applications of these techniques as this chapter demonstrates. This chapter presented how deep learning techniques, machine learning, and reinforcement learning can solve technical problems of welding processes. The stages developed by data mining processes can contribute in other areas to carry out an efficient investigation of welding processes.

3 DATA ANALYSIS AND MODELING TECHNIQUES OF WELDING PROCESSES: THE STATES-OF-ARTS AND METHODOLOGICAL PROPOSALS

The analysis of the State-of-the-art of welding processes will be developed according to some stages mentioned in the section of Data Mining Processes (see chapter 2). This allows a better understanding of possible applications of the new techniques and the most utilized at each stage of welding investigations.

3.1 SENSORS

Several sensors have been applied in welding processes to obtain information as can be seen in Figure 3.1. The utilization of infrared vision techniques has been applied in researches on welding processes as shown in (Fidali and Jamrozik 2013, Sreedhar et al. 2012, Bagavathiappan et al. 2013, Coniglio et al. 2016). Thus, one of the problems of this technique is that the environment where it is applied can interfere with the precision of the data obtained from the process. This may be due to the own heat emission of the technologies been utilized. Another type of sensor with great potential in welding processes is the sound sensor.



Figure 3.1: Diagram of sensors

Source: (Bestard 2017)

3.1.1 Vision sensor

Vision sensor is largely utilized in welding process to analyze welding pool (Xu et al. 2012, Liu, Huang and Zhang 2015), welding arc (Fei Gao and Qinglan Chen 2015, Ogawa 2011) and weld bead geometry (Günther et al. 2014, Günther et al. 2016). The high light generates by arc becomes hard the images obtentions. In this case, some techniques are utilized. One of them is utilized by (Chen and Farson 2010). This author made effective monitoring and control of the hybrid laser/gas metal arc welding process quality with an economical sensor system. A coaxial vision system for the hybrid process monitoring was integrated from a relatively inexpensive industrial vision system and a personal computer (PC). Other visualization technique is *Shadowgraphy*, applied in (Ramos, Carvalho and Absi Alfaro 2013) and (Siewert et al. 2014). This is based on obtaining the shadow of the process through the utilization of a laser source. With high-speed illumination laser, (Ma, Li and Chen 2017) obtained great quality images. This technique is new but it needs a laser with more potential than the traditional Shadowgraphy technique.

3.2 IMAGES PROCESSING

Some papers defined their structured light image processing technology into pipeline welding automation projects. It develops a vision-based pipeline too, as shown in (Yue et al. 2009). Here, Weld image processing adopts the base theory, including the Laplacian of a Gaussian filter, the neighborhood mean filter, the largest variance threshold segmentation, and morphologic. Other works like (Xu et al. 2014) applies the main characteristics of the gray gradient. A new improved Canny edge detection algorithm was proposed to detect the weld-image edges and extract the seam and pool characteristic parameters. (Wu et al. 2014) drove to find out the optimal noise filtering algorithm. It made a comparison of three noise filters: Gaussian filter, Median filter, and Wiener filter with welding seam image captured from the CCD camera. As a result, this paper considers the Median filter to have a better enhancement effect than the other two filters. In classic image processing, it difficult to generalize a filter or algorithm because it depends on the conditions and characteristics of camera parameters and light. According (Redmon et al. 2016), deep learning techniques have efficient result in real-time executions, while (Zhu et al. 2014) and (Pachitariu et al. 2013) affirm that the classification is better. One example applied in welding process is (Günther et al. 2014, Günther et al. 2016). It utilizes the autoencoder deep learning technique to extract features of images process in laser welding. Hence, the deep autoencoder features yielded a lower classification error when utilized as input for two Support Vector Machine (SVM) classifiers. Not only focusing on welding arc analysis but with good results, (Hou et al. 2018) propose an automatic detection for weld defects in x-ray images. It is constructed based on a deep neural network and classification model which was trained and tested by the patches cropped from x-ray images. Consequently, the proposed model obtains a maximum classification accuracy rate of 91.84%. This was one more example of the potential of these techniques in welding area.

3.3 MODELING A WELDING PROCESS

Demand for quality products has led to the rapid advancement of today's manufacturing environments. Many techniques and methods are applied to correlate between process parameters and bead geometry. One of them was the Response Surface Methodology (RSM). It was applied by (Sen 2015), where he evaluated the correlations that occurred between double pulsed gas metal arc welding (DP-GMAW) process parameters and weld bead geometry. (Santhana Babu et al. 2016) have acquired good results with the same technique to predict and control the weld bead quality in GTAW process. The problem of this method is that researchers should find the equation called response surface by test and error which can be hard. Many theoretical models have been defined to determine the process that occurs in the welding arc including (Boutaghane et al. 2011). The main problem of these models was that they lose precision because it was tricky to obtain a formula that contains all the complexity of the processes as affirmed by (Dong et al. 2016).

Mathematical models based on machine learning techniques have better results in problems as complex as this one. In the same paper, (Dong et al. 2016) expresses the potential of these models. One of the well-known and utilized regression algorithms is the least-squares method. It was utilized in (Gao et al. 2011) to predict the seam position directly under strong disturbing influence from the arc light. (Li and Gao 2014) utilized a linear regression model between pool image centroid deviation and the weld based for visual weld deviation measurement in GTAW process. Another technique was Gaussian Process Regression(GP) utilized in (Dong et al. 2016) to predict the characteristic performance of an arc welding process in GTAW.

One interesting method utilized in (Feng et al. 2012), is Mahalanobis Distance Measurement. It was illustrated and employed to determine whether welding faults have occurred or not. The same method was utilized in 2017 by Khairul Muzaka in his work (Muzaka et al. 2017) on GMAW process to optimize welding current for A vertical-position welding. The worst thing with this method is that it only correlates in the function of one input. (Bai and Lubecki 2016) proposed an on-line analysis method based on Localized Minimum and Maximum (LMM) of the welding process stability, for a welding monitoring system. The problem of LMM is to show a simple function to quality measures, then not define the complexity of the system. And that is why this work is limited only to the short circuit transfer mode.

(Park and Kim 2017) proposed an SVM with bootstrap aggregating to improve the prediction accuracy on the noisy RSW data with computational efficiency. In this framework join other technique as Generalized Regressive Neural Networks (GRNN) and Genetic algorithms for optimization. This article demonstrates an increase in more complex computer science techniques for better analysis of welding processes. Even though the only way to know if all this is the best solution is by comparing with other techniques.

3.3.1 Artificial Neural network models (ANN)

Some researchers already had references to these algorithm's advantages. Like Bo Chen that utilized ANN and Dempster-Shafer evidence theory, in (Chen, Wang and Chen 2010), to predict the penetration status in GTAW process.

They have also been utilized for different purposes and in different welding processes like:

- In SAW process, to predicting weld bead geometry (Sarkar et al. 2016).
- In GMAW Cold Metal Transfer (CMT) process, to predicting weld bead geometry (Pavan Kumar et al. 2017).
- In GTAW process, it is predicted the angular distortion considering the weld bead geometry (Rong et al. 2016).
- In Girth Welded Pipes process, to predicting of residual stresses (Mathew et al. 2017).
- In Underwater Wet Welding Process, to predicting the weld seam's geometric parameters (Chen and Feng 2014).

ANNs have been mixed with other techniques to obtain better results. One example of this is shown in (You, Gao and Katayama 2015), where it utilizes ANN and Support Vector Machine (SVM) for monitoring a welding defect in a laser welding process. (Chen and Chen 2010) predict the penetration in GTAW process, but used different ANNs to process information from different sensors, and finally predictive fuzzy integral method. Another example is in (Rios-Cabrera et al. 2016), where ANN Fuzzy ARTMAP was utilized to predict bead width and height in GMAW process like monitoring task.

The increase in computational resources has allowed an increase in the complexity of ANN architectures. These are called Deep Neural Networks (DNN). Bit by bit, they begin to be applied in the welding process. One of them was utilized in (Keshmiri et al. 2015). The model is based on a four-hidden-layer neural network architecture to make a study of the estimation of weld bead parameters. This article mixed data from different welding processes. This is a risk for results analysis since different processes can have different outcomes with the same input parameters. (Rao, Srinivasa Rao and Deepak 2017) utilized the Generalized Regressive Neural Networks (GRNN) technique, for estimating and optimizing the vibratory assisted welding parameters, to produce quality welded joints. But in this case, it does not have a comparison with other algorithms. (Wu et al. 2017) wrote a paper addressing "t-stochastic neighbor embedding" and deep belief network (DBN), other DNN variant, to perform variable polarity plasma arc welding (VPPAW) process monitoring and penetration status identification. Thus experimental verification and comparisons show that the classification performance of DBN can reach 97.62%, which indicates DBN outperforms ANN and support vector machine (SVM) models. This reaffirms the good results offered by the learning models developed with these algorithms. This work did not have as an objective the use of DNN algorithms to analyze images and sound in real-time, which could have been very interesting in the research.

Despite being the most used technique in welding processes analysis, Figure 3.2 presents that the best results are not always obtained. This figure was created with the articles studied in this review, and it proves that, with a comparative analysis, better models can be identified with a lower computational cost.

3.3.2 Why is it necessary to test and compare different models?

As it has been expressed in the previous sections, there are new techniques to analyze complex systems. But they require expensive computational resources for their construction and sometimes for their execution. A comparison between models will allow knowing which model has better results and which model can be the most effective to be utilized. This effectivity is measured in function of problem necessity like it is shown in data mining (DM) methodologies and processes (Piatetsky 2014, Chambers, Doig and Stokes-Rees 2017).



Figure 3.2: comparison between ANNs and ANN variations

The best result, for weld deviation extraction and weld groove state in Rotating Arc Narrow Gap MAG welding (RANGMW), was obtained by a comparison between Support Vector Machine (SVM) and ANN model (Li et al. 2014). This showed that the SVM model's predictive ability was better than the ANN model because it adapts to the little sample problem and can avoid the local extreme. One comparison with a focus on time optimized was (Kumar et al. 2014). Here, it utilized an ANN and ANN with differential evolutionary algorithm (DEA) separately. The results obtained were closer to ANN, but the computational time of ANN using DEA was shorter than the other algorithm. In the article, (Escribano-García et al. 2014) Response Surface Methodology (RSM) was compared with regression models based on DM (linear regression (LR), isotonic regression (IR), GP, ANN, SVM and regression trees (RT)) to evaluate mechanical properties in GMAW process. The results showed that the regression models obtained with DM

Source: Produced by author

generally have poorer generalization capacity than the regression model obtained with RSM. Because DM techniques require a relatively large amount of data to obtain acceptable results. The article (Sumesh et al. 2015) compared decision trees (DT), ANN, fuzzy logic, SVM, and random forest techniques for weld quality monitoring in SMAW. It knows the importance of comparing data mining techniques. The most efficient technique was the random forest. This shows that not always the most complex techniques offer the best results. One of the few comparative analyses algorithms was shown in (Kumar et al. 2016). This paper explores the self-organizing maps(SOM) algorithm as a mechanism for performing unsupervised learning. It compared the performance, characteristics of various welding parameters. The SOM result was compared with the Probability Density Distributions (PDDs) obtained during statistical analysis. Finally, it is shown that, in addition to PDD, analysis of voltage and current data using the SOM technique can also be utilized to evaluate arc welding process. These studies demonstrate that there are other potential algorithms for step analysis in welding process. And it is necessary to evaluate and compare several of them to select the best upon in a real-time process.

Another comparison was done by (Wu et al. 2016). The article compared a prediction model of Plasma Arc Welding based on the Extreme Learning Machine (ELM) technique with ANN and SVN. This proposed model is faster and has better generalization performance. This potentiality is established by (Nandhitha 2016). In this case, he utilized Radial Basis Networks (RBN) and Generalized Regressive Neural Networks (GRNN) to torch current prediction in GTAW process. GRNN outperforms RBN in predicting the torch current deviation with 98.95 % accuracy. (Kim, Park and Sohmshetty 2017) discusses the Resistance Spot Welding (RSW) process. He examines the prediction performance of k-nearest neighbor (kNN) and GRNN. The results indicate that using a smaller k on properly-inconsistent data increases the prediction performance measured by mean acceptable error (MACE).

Another quality welding article was (Wan et al. 2017). A different neural network model was proposed for weld quality prediction in large scale RSW process. In the research, a simple ANN was more proper in failure load estimation. The probabilistic neural network model was more appropriate to be applied in quality level classification. One of the few articles with DM techniques mention in welding process was (Huang et al. 2017). This is an investigation of porosity in pulsed Gas tungsten arc welding (P-GTAW) of aluminum alloys based on spectral and x-ray image analyses. This made spectral analyses based on DM and empirical mode decomposition (EMD) were proposed to detect porosity. (Petković 2017) predicted a laser welding quality by training data for the computational intelligence methodologies and support vector regression(SVR). Support Vector Regression is a novel variant of Support Vector Machine usually for regression tasks. This article made a comparison between SVR, ANN, and GP. It is another example that, in specific problems, less complex algorithms can offer better results. It defines which of the techniques is most effective for solving the problem. Moreover, it helps in the effectiveness of a future process of intelligent control.

Table 3.1 presents some articles that were based on quality monitoring of welding processes. The preparation column defines the processing technique of data obtained by the sensors. The classic value represents welding processes that do not use the newest techniques of image processing and
DL for those who use them. The modeling column defines the algorithms used in the specific article. The online column defines whether the proposed model was executed in real-time. The compare column defines whether this paper made a comparison between several algorithms. When there is a comparison, the first model before the comma is of the best quality result. As all tables show (Table 3.1, Table 3.2 and Table 3.3), the best algorithm does not always be the same.

author	year	welding process	sensors	Preparation	Modeling	Online	compare
(Saini 1998)	1998	GMAW	sound	classic	no	yes	no
(Yue et al. 2009)	2009	no speak	visual	classic	theorical model	no	no
(Chen and Farson 2010)	2010	LBW/GMAW	visual	classic	no	yes	no
(Horvat et al. 2011)	2011	GMAW	sound	classic	no	yes	no
(Gao et al. 2011)	2011	GTAW	visual	classic	LR-ANN	no	no
(Feng et al. 2012)	2012	GMAW	standard	classic	MDM	yes	no
(Kalaichelvi, Karthikeyan and Sivakumar 2013)	2013	GMAW	standard	classic	GA-Fuzzy	yes	no
(Fidali and Jamrozik 2013)	2013	GMAW	infrared	classic	statistical analysis	yes	no
(Sreedhar et al. 2012)	2013	GTAW	infrared	classic	statistical analysis	yes	no
(Kumar, Anand and Srivastava 2014)	2014	no speak	visual	classic	ANN	no	no
(Kumar et al. 2014)	2014	GMAW	visual	classic	ANN, ANN-DEA	yes	yes
(You, Gao and Katayama 2015)	2015	Laser welding	Spectrometer	classic	FFANN-SVM	yes	no
(Sumesh et al. 2015)	2015	SMAW	sound	classic	some DM (RF)	yes	yes
(Baraka, Panoutsos and Cater 2015)	2015	FSW	standard	classic	ANN-Fuzzy, ANN	yes	yes
(Kumar et al. 2016)	2016	SMAW	standard	classic	PDDs, SOM	no	yes
(Muzaka et al. 2017)	2016	GMAW	standard	classic	MDM	yes	no
(Bai and Lubecki 2016)	2016	GMAW	standard	classic	LMM	yes	no
(Park and Kim 2017)	2017	RSW	standard	classic	GRNN-SVM	yes	no
(Wan et al. 2017)	2017	LSRSW	standard	classic	ANN(BP), ANN(Prob)	yes	yes
(Huang et al. 2017)	2017	P-GTAW	visual	classic	DM, EMD	no	yes
(Petković 2017)	2017	Laser welding	multiples	classic	SVM, ANN, GP	yes	yes
(Muniategui et al. 2017)	2017	RSW	visual	DL, classic	fuzzy	yes	yes

Table 3.1: Table articles with quality objective

Table 3.2: Table articles with prediction objective

author	year	welding process	sensors	Preparation	Modeling	Online	compare
(Chen, Wang and Chen 2010)	2009	GTAW	multiples	classic	ANN-DS	no	no
(Chen and Chen 2010)	2010	GTAW	multiples	classic	ANN-Fuzzy	no	no
(Seyyedian Choobi, Haghpanahi and Sedighi 2012)	2012	Butt welding	standard	classic	ANN	yes	no
(Chiumenti et al. 2013)	2013	FSW	standard	classic	math model	no	no
(Li and Gao 2014)	2014	GTAW	visual	classic	LR	no	no
(Chen and Feng 2014)	2014	UWW	visual	classic	ANN	yes	no
(Escribano-García et al. 2014)	2014	GMAW	standard	classic	RSM, some DM	yes	yes
(Li et al. 2014)	2014	RANGMW	visual	classic	SVM, ANN	yes	yes
(Sen 2015)	2015	DP-GMAW	standard	classic	Taguchi-RSM	no	no
(Keshmiri et al. 2015)	2015	GMAW, GTAW	standard	classic	DNN	yes	no
(Nandhitha 2016)	2016	GTAW	thermografy	classic	ELM, RBN, GRNN	yes	yes
(Wu et al. 2016)	2016	VPPAW	sound	classic	ELM, ANN, SVM	yes	yes
(Lv et al. 2016)	2016	GTAW	sound	classic	BP-Adaboost	yes	yes
(Dong et al. 2016)	2016	GTAW	standard	classic	GPR	yes	no
(Kim, Park and Sohmshetty 2017)	2016	RSW	standard	classic	kNN, GRNN	yes	yes
(Sarkar et al. 2016)	2016	SAW	standard	classic	MRA, ANN	yes	yes
(Rong et al. 2016)	2016	GTAW	standard	classic	ANN	yes	no
(Rios-Cabrera et al. 2016)	2016	GMAW	visual	classic	ANN-Fuzzy-ARTMAP	yes	no
(Aviles-Viñas, Rios-Cabrera and Lopez-Juarez 2016)	2016	GMAW	visual	classic	ANN-Fuzzy	yes	no
(Pavan Kumar et al. 2017)	2017	GMAW CMT	standard	classic	ANN	yes	no
(Mathew et al. 2017)	2017	Butt welding	standard	classic	ANN	yes	no
(Wu et al. 2017)	2017	VPPAW	visual, sound	classic	t-SNE and DBN	no	no
(Wan et al. 2017)	2017	RSW	standard	classic	ANN, LR	no	yes

Table 3.1 and Table 3.2 also show a scarcity of comparative analysis, and little application of these machine learning techniques, especially in the GMAW process. Most of these papers do not take advantage of deep learning techniques in image processing. This highlights the innovative potential of applying these techniques in welding processes.

3.4 INTELLIGENT CONTROL

The intelligent control approach offers interesting perspectives since it can provide methodologies that allow performing some of the tasks typically performed by humans automatically (Santos 2011). This combines with data mining models.

One intelligent control tendency is to utilize fuzzy methods with an ANN model. An example of this is shown in (Chen, Wang and Ma 2010). It predicted dynamical characteristics of the weld pool during robotic welding in GTAW process. In (Hailin et al. 2012), GMAW pipe-line welding is shown to improve the welding quality. (Cruz, Torres and Alfaro 2015) is another example of modeling and control in GMAW process. Other fuzzy methods examples but a different technique was shown in (Sharma, Maheshwari and Rathee 2016). This article proposes a response to a fuzzy logic approach with surface methodology (RSM), demonstrating that, any model obtained from a welding process can be integrated into a control system as long as it meets time demands.

An emerging control system was used by (Günther et al. 2016) for Laser welding Control. This technique is called reinforcement learning (RL) and it is a branch of machine learning and artificial intelligence. It is focused on goal-directed learning and decision making (Sutton and Barto 2017). Integral to RL approach are methods for learning expectations of future observations from samples of experience (prediction learning), and using samples of experience to affect policy change (control learning) (Günther et al. 2016). Control learning can be an optimization-based method like a q-learning algorithm. Also, it can be used to solve optimal control problems (Li et al. 2018). (Günther et al. 2016) used reinforcement learning to acquire generalized predictions for use as inputs to a laser welding system. This makes this work an important contribution to welding process engineering. Likewise, RL is a new technique open now in welding process with noble success in other areas like it shows (Chincoli and Liotta 2018, Ramanathan, Mangla and Satpathy 2018, Yin, Yu and Zhou 2018).

Table 3.3 presents a summary of some welding area articles that apply intelligent control techniques. In addition to showing the same pattern as the previous Table 3.1 and Table 3.2, it exhibits a high application of fuzzy algorithms for process control. For this reason, the application of reinforcement learning in a GMAW process can be considered innovative in this research.

author	year	welding process	sensors	Preparation	Modeling	Online	compare
(Chen et al. 2000)	2000	P-GTAW	doble-visual	classic	ANN-learning Control	yes	yes
(Chen, Wang and Ma 2010)	2009	GTAW	visual	classic	ANN-Fuzzy	yes	no
(Malviya and Pratihar 2011)	2011	GMAW	standard	classic	ANN-PSO	yes	no
(Hailin et al. 2012)	2012	GMAW	visual	classic	ANN-Fuzzy	yes	no
(Wang 2014)	2014	GMAW	visual	classic	ANN-Fuzzy	yes	no
(Cruz, Torres and Alfaro 2015)	2015	GMAW	visual	classic	ANN-Fuzzy	yes	no
(Santhana Babu et al. 2016)	2016	GTAW	standard	classic	RSM	yes	no
(Günther et al. 2016)	2016	Laser welding	visual	DL	DL-RL	yes	no
(Santhana Babu et al. 2016)	2016	GTAW	standard	classic	RSM	yes	no
(Sharma, Maheshwari and Rathee 2016)	2016	SAW	standard	classic	RSM-Fuzzy	yes	no
(Azadi Moghaddam, Golmezergi and Kolahan 2016)	2016	GMAW	visual	classic	ANN-PSO	yes	no
(Lv et al. 2017)	2017	GTAW	sound	classic	ANN	yes	no
(Rao, Srinivasa Rao and Deepak 2017)	2017	Vibratory Welding	standard	classic	GRNN	yes	no
(Hu, Huang and Zeng 2017)	2017	GMAW	standard	classic	math-model-Fuzzy	yes	no

Table 3.3: Table articles with control objective

3.5 FUTURE PERSPECTIVE

These data analysis techniques based on learning, as appear in this article, are not yet widespread in welding process area. A bibliometric analysis among the authors studied in this research presents a very little relationship between them. Figure 3.3 exposes this affirmation.



Figure 3.3: Bibliometric analysis: Authors interrelationship

Source: Produced by author

In this study, all the articles referenced in this research were analyzed. Through the interrelationship between authors is validated with their participation in the same publication. Consequently, in the articles analyzed, this interrelation did not go beyond level 2 (the authors only participated jointly in 2 publications). In addition, this level 2 tends to be among the same researchers. All these affirm, the small dimensions of the authors' clouds (articles with welding process and new data analysis techniques) and their small relationships (Author interaction by publication), show little maturity in the interrelation of these areas. The cases that show a greater cloud of relationship, is due to the participation of many authors in the same publication.

Some of the works demonstrate a small approximation between areas, fulfilling the interdisciplinarity that industry 4.0 advocates. Achieving this interdisciplinarity implies new study processes. For this reason, it defines new methodologies that unify the potential of these two areas. The needs of the modern world are going to make this happen in a short time. The new data analysis conception in welding processes area will be an acceleration in obtaining new and better models more efficient predictions and controls.

3.6 METHODOLOGIC DIAGRAMS

After the previous analysis on algorithms and techniques used for information acquisition, prediction, and control in welding process. It can be observed that there are computer science techniques little experienced in welding processes, and even more in GMAW process. This research proposes the development of two methodologic diagrams with these techniques. Diagram 1 (Drop-Volume Methodology), Figure 3.4, will be analyzed with some GMAW arc parameters like unmelted wire length, drop volume, and melted wire volume. Diagram 2 (Drop-Frequency Methodology), Figure 3.5, will be analyzed with by GMAW arc parameters of drop frequency and short circuit frequency.





Source: Produced by author

Drop Volume Methodologic will be analyzed with some GMAW arc parameters like unmelted wire length, drop volume, and melted wire volume. This process has more parameters, which can allow you more precision in the results, like show in (Thompson Martínez et al. 2021).





Source: Produced by author

The drop frequency methodology accumulates the data processing values for a time interval and then applies the predictive model. It will allow better computational performance than the drop volume methodology. Thereby, both methodologic diagram will be developed in next chapters. All stages will be compared and evaluated to define better process. Both have the same experiments and process, they only differ in the parameters obtained in the image processing. The diagrams were developed with Integration Definition for Function Modeling (IDEF0) (United States Air Force commissioned). Each box represents the processes that were developed. Horizontal arrows mean inputs or outputs parameters. While, vertical bottom arrows mean the mechanism needed to executed the function. Vertical top arrows mean the control system.

- The first step is focused on the parameters acquisition of experimental from the GMAW process with the help of power supply and welding table. This require the develop of experimental plan.
- The second step aims to capture images of what is happening in the GMAW welding arc, based on the parameters provided by initial step. The mechanism used to develop this objective was a high speed camera.
- The third step aims to develop a deep learning model to detect the droplet detachment and the occurrence of short circuit. This model will allow the calculation of unmelted wire length and melted wire volume for drop volume methodology, short circuit and droplet frequency for the drop frequency methodology.
- The fourth step aims to develop a predictive model, using machine learning techniques, based on the parameters generated in the previous step and the input parameters of the GMAW process. Predictive model will output parameters of the weld bead geometry.
- The fifth step aims to develop intelligent control, using a reinforcement learning model. This model would have the ability to generate which are the most acceptable control parameters to meet a requirement of weld bead geometry in GMAW process.

The developed diagrams can also be analyzed as the process flow of the system prototype, once the models are satisfactorily validated. For these validations, the strategies and propositions of the data mining process will be taken into account. As a consequence, the results will allow an analysis and comparison of the two proposed methodologies.

3.7 CHAPTER CONSIDERATIONS

In this chapter, we carried out an analysis of several articles about the welding process. It allowed determining for each data mining stage how it is possible to optimize the results to obtain a good result of process analysis. Several analysis algorithms of the welding process were shown, and it was demonstrated that the comparison between them can make the process analysis more efficient and less expensive. The potential of learning-based techniques is described because

computational resources are becoming cheaper and welding information can be obtained more and more quality information. All these premises aligned with the so-called industry 4.0 where a set of technologies that allow a fusion of the physical and digital world, to create more intelligent systems.

4 EQUIPMENT, MATERIALS AND DATA ACQUISITION OF THE EXPERIMENTS

This chapter explains the equipment and materials used during the research development. The strategy developed for data-acquisition through the experiments is also shown. The location of these pieces of equipment is in the Automation and Control Group laboratory (GRACO) of the University of Brasilia (UnB). They are:

- Welding supply: is a fully digitized microprocessor controlled inverter power sources. The modular design and potential for system add-ons ensure a high degree of flexibility. The device can be adapted to any specific situation (Fronius 2012).
- Welding Supplied Interface: A communication system was used between the welding supply and the computer, which was developed by (Moncayo Torres 2013, Giron Cruz 2014) in GRACO. It was developed to control the welding supply parameters (voltage and wire-rate).
- Welding Table: The welding table used was developed by the research work (Díaz Franco 2008) in GRACO, it is a linear table with one-dimensional movement. it has a control system translation speed (welding speed) as shown in Figure 4.1.
- **High-speed camera:** the camera provides full mega pixel resolution, images at frame rates up to 3,000 frames per second (fps), 512 x 512 pixels. This camera has e Gigabit Ethernet and Optical interfaces available, as it is shown in figure 4.1. In these experiments, a frame rate of 1000 fps and resolution of 1024 px were used.

Figure 4.1: Welding table, camera and laser



Source: Produced by author

4.0.1 Materials

The materials used in this study are electrode wire copper with 1,2 mm of diameter. The base material is a 1020 steel in flat sheet format, dimensions 6,35 mm thick, and 300 mm x 40 mm long and width respectively. Shielding gas used 96% argon and 4% carbon dioxide.

4.0.2 Visualization technique

The technique used to visualize the welding arc is Shadowgraphy, it was also used in (Mota et al. 2013, Ramos, Carvalho and Absi Alfaro 2013, Siewert et al. 2014). The term Shadowgraphy has been used to refer to a projected shadow of several elements in the welding region (torch, electrode, droplets, weld bead and plate) over a flat surface, technique also known as Back-lighting (Figure 4.2). As the arc light is too intensive and it irradiates to all directions, its use is not possible to obtain the shadow (Balsamo et al. 2000). This is achieved with laser of 633 nm, diverging lens, converging lens and bandpass filter of 500 nm - 700 nm as it is shown in figure 4.1. To obtain the images, it was necessary to align the laser with a high-speed camera.

Figure 4.2: The principle of Shadowgraphy Back-lighting applied to welding



Source: (Balsamo et al. 2000)

4.1 PLANNING OF EXPERIMENTS

The application used to configure the parameters of the welding supply and welding table. It was developed by the investigations (Moncayo Torres 2013, Giron Cruz 2014) and modifications of the doctoral student Jairo Muñoz Chavez. For this investigation, 3 data-acquisition experiments are developed.

4.1.1 Experiments 1 and 2

Data experiments 1 and 2 were obtained with factorial design of Central Composite Design (NIST), this is because it generates a reasonable data distribution with long possible distribution in the data set. The objective here is to obtain the greatest possible variety of data for the predictive model creation. In Table 4.1 and Table 4.2, it is shown, the experimental input parameters by time.

time(s)	wire rate(m/min)	voltage(v)	welding speed(mm/s)
0	5.5	20	8
3	5.5	20	8
5	7.5	20	8
7	5.5	29	8
9	7.5	29	8
11	5.5	20	12
13	7.5	20	12
15	5.5	29	12
17	7.5	29	12

Table 4.1: Experiment 1: input parameters by time

4.1.1.1 Results of experiments 1 and 2

As a result of the aforementioned experiments, the following were obtained: 24 578 and 24 577 welding arc images of the experiment 1 and experiment 2 respectively. Figure 4.5

time(s)	wire rate(m/min)	voltage(v)	welding speed(mm/s)
0	4.8	24.5	10
3	4.8	24.5	10
5	8.2	24.5	10
7	6.5	17	10
9	6.5	32	10
11	6.5	24.5	6.6
13	6.5	24.5	13.4
15	6.5	24.5	10
17	6.5	24.5	10

Table 4.2: Experiment 2: input parameters by time

and Figure 4.6 are two examples of welding arc images obtain by high-speed camera and shadowgraphy technique.



Figure 4.3: Experiment 1: Example of Welding arc images No. 9608

Source: Produced by author

Figure 4.4: Experiment 2: Example of Welding arc image No. 16458



Source: Produced by author

4.1.1.2 Parameters of weld bead geometry

The weld bead geometries ((width, depth, and height)) were obtained from the macrographic analysis. The macrographic analysis was made in a longitudinal direction, this is in the direction of the torch movement. In these cases, it is taken the maximum value in each measurement point, as expressed in (Bestard 2017, Alvarez Bestard and Absi Alfaro 2018, Bestard et al. 2018). To obtain weld bead geometry data was necessary polished and etched using 2.5% nital solution to display the weld bead penetration. Figure 4.5 and Figure 4.6 show an example of the resulting pieces. Thus, the dimensions were obtained using an image processing algorithm. This calculates the geometric parameters from the horizontal line to find the parameters color by the vertical line. The values were transformed to the corresponding scale.



4.1.1.3 Output parameters

The dimensions of the weld bead geometry were obtained using an image processing algorithm. They were painted on the image, the dimensions corresponding to weld bead geometry like shown in Figure 4.5 and Figure 4.6. The algorithm is shown in annex A and diagram flow in Figure 4.7. It was applied to convert the values in respective millimeters dimensions. Similarly, for this algorithm is necessary to know the pixel relationships with millimeters dimensions, the coordinates of the beginning and the end of the process, and the colors with which they were drawn. Figure 4.8, Figure 4.9, and Figure 4.10 are the weld bead geometries dimension of experiment 1. In addition, Figure 4.11, Figure 4.12, and Figure 4.13 are the weld bead geometries dimension of experiment 2.

Figure 4.7: Sequence diagram of weld bead geometry calc



(*) The equation for depth changes to $y = y_i - 1$ Source: Produced by author



Source: Produced by author

Figure 4.9: Experiment 1: Height geometry vs time



Source: Produced by author

Figure 4.10: Experiment 1: Depth geometry vs time



Source: Produced by author



Source: Produced by author

Figure 4.12: Experiment 2: Height geometry vs time



Source: Produced by author

Figure 4.13: Experiment 2: Depth geometry vs time



Source: Produced by author

4.1.2 Experiments 3

In the experiment, wire-rate speed and welding speed were constants. Voltage value is increased 1,5 v every 2 s as shown in weld bead Table 4.3. The objective of this experiment was to gather images of the three main transfer modes and to test the deep learning model proposal.

time(s)	wire rate(m/min)	voltage(v)	welding speed(mm/s)
0	6.8	19	9
2.5	6.8	19	9
4.5	6.8	20.5	9
6.5	6.8	22	9
8.5	6.8	23.5	9
10.5	6.8	25	9
12.5	6.8	26.5	9
14.5	6.8	28	9
16.5	6.8	29.5	9

Table 4.3: Experiment 3: input parameters by time

4.1.2.1 Results of experiments 3





Source: Produced by author

Figure 4.15: Experiment 3: Weld bead geometry



4.2 CHAPTER CONSIDERATIONS

The planned experiments allowed compiling the necessary data in the analyses developed in the following chapters. Therefore, images of the arc from the three experiments, as well as the input, and output parameters, were collected. The final dataset preparation was carried out using time as a reference and considering the response time of the equipment used and the process.

5 RESULTS OF THE TWO METHODOLOGIES PROPOSALS FOR WELD BEAD GEOMETRY MODELING IN GMAW PROCESS

In this chapter, the two methodological diagrams proposed in Chapter 3 will be developed. The objective of this chapter is to validate the results of the two methodological diagrams and to carry out a comparative analysis between them. The data collected were those gathered through the experimental designs shown in the previous chapter (Chapter 4). Thus, these data are conformed by input parameters (wire-rate speed, voltage, welding speed, and process time) of the process and by arc images obtained. The Welding arc image token contains information like unmelted wire length and drops detachment. Others parameters can be taken into the analysis of welding arc image are short circuit frequency and drop frequency. These data were correlated with the parameters of the weld bead geometry for the development of the models.

5.1 DATA PREPARATION

The data preparation phase covers all the activities required to construct the final dataset from the initial raw data. Data Preparation tasks are likely to be performed repeatedly and not in any prescribed order (Chapman et al. 2000, Marbán, Mariscal and Segovia 2009).

5.1.1 Short circuit and drop detection

For application molten or drop volume equation shown by (Choi et al. 2001). It is necessary to detect drop detachment or short circuit in a GMAW process by image sequence. In this process, the background subtraction techniques help.

Background subtraction is a widely used approach for detecting moving objects in videos from a static camera (Bouwmans et al. 2014). Frame differencing is the simple way to detect movement like is showed by (Bouwmans et al. 2014, Bouwmans, Aybat and ZAHZAH 2016). The frame differencing equation used in this work is:

$$F(t) = I_{t-2} - \frac{I_t}{3}$$
(5.1)

Where I_t is the image on same time t.

The division by 3 aims to keep visible image elements because the spatial patterns conformed by the common elements and the movement produced allows a better classification analysis and error reduction. The final pattern is showed in figures 5.1 and 5.2.





As shown in Figures 5.1 and 5.2, F(I) is the result of applying the equation 5.1. As a result of this, different patterns can be observed in short circuit and drop detachment, figure 5.1 compares F(I) result with figure 5.2. This pattern differences can be detected with deep learning technique of Convolutional Neural Network.





5.1.1.1 Convolutional neural network model

The best result in image classification is a convolutional neural network (CNN). CNN is a deep learning technique designed to work with two dimension patterns. It was the first successful hierarchical learning algorithm. It is a topology trainer that balances the spatial relationship to reduce the number of parameters that must be learned. Thus, virtually all feed-forward back propagation formation better (Arel, Rose and Karnowski 2010). The CNN architecture selection is ResNet (He et al. 2015, Szegedy et al. 2016).The code used is shown in annex B.

The architecture was defined with 18 hidden layers and adam optimization. Adam is an adaptive learning rate optimization algorithm that is been designed specifically for training deep neural networks (Kingma and Ba 2015). Deep learning models are model by learning. For this reason, they need groups of train images, test images, and validation images. In this process, it is necessary, a manual division of images into classes. These classes are the categories for the model. In this research the classes are:

- Short circuit and drop-detachment images (SCDD), F(t) of figure 5.1.
- No short circuit and no drop-detachment images (no SCDD), F(t) of figure 5.2.

This Table 5.1 show images distribution.

	SCDD	no SCDD
Train	618	2783
Test	100	408
Validation	42	212

Table 5.1: Images distribution by class and process

The model obtained shows 97,24% of correct test images group classification and loss value of 0,10.Figure 5.3 show the ResNet-model confusion matrix analyzed with validation values. It is a performance measurement for a machine learning classification problem where output can be in two or more classes. In this case, it means that 206 no SCDD and 39 SCDD were detected correctly. Only 7 no SCDD and 2 SCDD were incorrectly classified, which shows an excellent proportion in the classification process and excellent detection of the short circuit, drop detachment, and the contrary cases.

Figure 5.3: RestNet-model confusion matrix



Source: Produced by author

Up to this step, functionalities common to the two methodologies shown in the research have been developed. The following stages develop topics that can be specific to some of the defined methodologies.

5.1.2 Drop Volume Methodology: calculate volume

Based on the obtained parameters, an analysis was made in which were used the volume calculation methods. Due to the wide usage of GMAW in the industry, numerous models have been presented to study the droplet transition process. Typical approaches include:

- Static Force Balance Model: it predicts drop detachment by comparing the surface tension of the drop with the external forces exerted on the drop (Version 2004).
- Pinch Instability Theory(PIT): is a detachment criterion can be derived, that does not rely on the balance of axial forces, but rather relies on radial forces (Version 2004, Wang, Lü and Jing 2016).
- Mass-Spring Model: this has lead to simple calculation as well as reflected oscillation during droplet growth as it is shown in (Wang, Lü and Jing 2016).
- Volume-of-Fluid (VOF) method: VOF method is based on the magnetohydrodynamic analysis (Zhao and Chung 2018, Murphy 2013, Wang, Huang and Zhang 2004).

All these methods use equations in which they involve a spherical analysis of drop. But in (Choi et al. 2001), based on VOF and PIT. It showed a good analysis that when the drop detaches. The entire drop volume is assumed to be ejected from the wire tip. A drop detachment detection by image processing would allow the use of simple equations that are proposed in this article. The equations 5.2 and 5.3 was token of him.

$$\frac{dl_{es}}{dt} = v_f - v_m \tag{5.2}$$

$$\frac{dV_d}{dt} = \left(\frac{\pi * D_e^2}{4}\right) * v_m$$
(5.3)

Where l_{es} represents unmelted wire extension, v_f the wire-feed rate, v_m the wire-melting rate. Where V_d the attached drop volume, and D_e the wire diameter. Uniting the two equations is obtain as a result the equation 5.4

$$V_d = \left(\frac{\pi * D_e^2}{4}\right) \int (v_f dt - l_{es})$$
(5.4)

The unmelted wire length (l_{es}) necessary for volume calculation is obtained with classical images processing. The method used is shown in annex C, and the algorithm has the flow shows in Figure 5.4. The arc image is binarized and a pixel count is made vertically from the center of the wire until a change in pixel colors occurs. This process is showed in figure 5.5.



Source: Produced by author

Figure 5.5: Diagram of unmelted wire length calculation



Source: Produced by author

5.1.2.1 Result of volume calculation

Figure 5.6 belonging to experiment number 3 shows a distribution of the largest volumes and lower volume frequencies in the firsts seconds of GMAW process. In the middle, it shows a decrease in the unmelted wire size, a greater frequency, and a decrease in the molten wire volume. It characterizes the globular metal transfer mode in this process. In the last seconds, it shows the spray transfer mode through the reduction of the unmelted wire length and increases in drops frequency. This demonstrates a valid characterization of the process with these techniques.



Figure 5.6: Volume distribution on unmelted wire length vs time

Source: Produced by author

5.1.2.2 Volume dataset characteristics

After calculation of the molten volume(or drop volume) and unmelted wire length. It was started the correlation of all data on one dataset. This generated a dataset of 2170 records, taking experiments 1 and 2 into account, figure 5.2 shows a summary of dataset formed. The column *main* represents average values, column *min* and *max* are minimum and maximum values of each parameters. Column *std* is standard deviation over requested parameters. The values of standard deviation validate that there is a great diversity of values in the range of maximum and minimum specified.

	main	std	min	max
$wire_v(m/min)$	6.38	1.06	4.80	8.20
voltage(volt)	26.09	3.63	17.00	32.00
$weld_v(mm/s)$	9.99	1.90	6.20	13.40
$wire_l(mm)$	13.78	1.12	9.00	15.00
$volume_{(mm^3)}$	0.74	0.98	$5.6 * 10^{-5}$	10.54
$depth_{(mm)}$	1.95	0.89	0.00	3.62
$height_{(mm)}$	1.77	0.86	0.00	3.33
width(mm)	5.94	1.28	2.67	8.21

Table 5.2: Volume dataset

5.1.3 Frequency dataset characteristics

With the parameter dataset generated in the previous section, the frequency dataset was generated removing some fields, like volume and unmelted wire length. Table 5.3 shows a summary of dataset formed. The row dropfreq represent short circuit and drop frequency detected on 50ms. It has 585 records.

	main	std	min	max
dropfreq(unit)	3.68	3.82	1.00	24.00
$wire_v(m/min)$	6.42	1.05	4.80	8.20
voltage(volt)	25.17	4.09	17.00	32.00
$weld_v(mm/s)$	10.06	1.90	6.20	13.40
$depth_{(mm)}$	1.97	0.86	0.00	3.62
$height_{(mm)}$	1.97	0.74	0.00	3.33
width(mm)	5.91	1.29	2.67	8.21

Table 5.3: Frequency dataset

The common parameters have similar behavior, which allows these two methodologies to be compared. The differences in the drop frequency dataset are related to the sampling frequency. This decreases the parameters for analysis.

5.2 PREDICTIVE MODEL FOR GMAW PROCESS

In the construction of this model, the scikit-learn library was used (INRIA). This library contains several machine learning algorithms that allow making a comparison between them. The selection of algorithms is based on those with good results in prediction tasks. They were:

• Lasso regression: is a type of linear regression that uses shrinkage. Shrinkage is where data values are shrunk towards a central point, like the mean. The lasso procedure encourages simple, sparse models (i.e. models with fewer parameters). (Tibshirani 1996, Friedman, Hastie and Tibshirani 2010).

Meta-parameters selected: alpha = 1.802e - 01, $fit_intercept = True$, normalize =

 $False, precompute = False, copy_X = True, max_iter = 1000, tol = 0.0001, warm_start = False, positive = False, random_state = None, selection =' cyclic'.$

• **Ridge regression**: is a technique for analyzing multiple regression data that suffer from multicollinearity (Fearn 2013, Breheny 2016). Ridge regression is like least squares but shrinks the estimated coefficients towards zero (Tibshirani 2013).

Meta-parameters selected: alpha = 1.802e - 01, $fit_intercept = True$, normalize = False, $copy_X = True$, $max_iter = None$, tol = 0.001, solver = 'auto', $random_state = None$.

• **Bayesian regression**: Bayesian regression is one of the techniques widely used since 1960s as shown in (Zellner and Chetty 1965, Currin et al. 1991, Sahu, Dey and Branco 2003, Chan and Vasconcelos 2012). It is based on probabilistic Bayes' Theorem with excellent results in classification and prediction task as demonstrated by (Pawlak 2002, Rudner and Liang 2002, McNamara, Green and Olsson 2006).

Meta-parameters selected: $n_i ter = 300, tol = 0.001, alpha_1 = 1e - 06, alpha_2 = 1e - 06, lambda_1 = 1e - 06, lambda_2 = 1e - 06, alpha_init = None, lambda_init = None, compute_score = False, fit_intercept = True, normalize = False, copy_X = True, verbose = False.$

• Support vector machines (SVMs): are a set of supervised learning methods used for classification, regression, and anomaly detection. They are effective in high dimensional spaces and still effective in cases where a number of dimensions are greater than the number of samples. Different Kernel functions can be specified for the decision function. Common kernels are provided, but it is also possible to specify custom kernels (Cortes and Vapnik 1995, Smola, Smola and Schölkopf 2004).

Meta-parameters selected: $gamma =' scale', coef 0 = 0.0, tol = 0.001, C = 1.0, epsilon = 0.1, shrinking = True, cache_size = 200, verbose = False, max_iter = -1.$

These algorithms, compared with ANN are less expensive in computational training process. They need less time for the development and definition of the learning hyperparameters. All these allow a reduction in research time, whether any of them offers a satisfactory result. Then it possibly discarded the use of neural networks based on the previous arguments. All this allows a reduction in research time if any of them show good results like it shows in table 5.4. The technique column is some machine learning techniques used in this research, RMSE is root mean square error. The grade represents the function grade for algorithms with polynomial functions.

5.2.1 Result of volume dataset

The best performance was obtained by SVM with radial basis function (RBF) kernel as can be seen in table 5.4. The RMSE of this algorithm was 0.27. The code used is shown in annex D.

technique	RMSE	grade
Lasso regression	0.30	5
Ridge regression	0.36	3
Bayesian regression	0.31	4
SVM polinomial	0.67	1
SVM-RBF	0.27	-

Table 5.4: Minimus error calculation of each models(Volume dataset)

5.2.2 Result of frequency dataset

The best performance was obtained by SVM with Radial Basis Function (RBF) kernel too, like it shows in table 5.5.

technique	RMSE	grade
Lasso regression	0.61	3
Ridge regression	0.39	3
Bayesian regression	0.40	3
SVM polinomial	0.89	1
SVM-RBF	0.33	-

Table 5.5: Minimum error calculation of each models(Frequency dataset)

The margin error with the volume dataset is not very large. Therefore, based on better performance, the SVM-RBF model of frequency dataset could be applied.

5.3 SIMULATION OF WELD BEAD GEOMETRY FOR GMAW PROCESS

The mathematical model was obtained in the previous section which characterizes the relation of the selected input parameters and the weld bead geometry on GMAW process. As defined in the section on intelligent modeling (chapter 2), the modeling can be defined as a reinforcement learning technique with a q-learning algorithm. This technique was analyzed in chapter 3. In the q-learning algorithm, it is necessary to define the states, actions, and rewards as is shown in figure 2.12 of chapter 2.

In this experiment, the state are going to be the vector of 'height', 'width', 'depth' values. This is because these are the parameters to be controlled and the interrelation between them makes it impossible to control one without modified the others. The actions can be increased, decrease or maintain the same for each value of vector 'voltage', 'wire-rate', 'welding velocity'. Based on the experience and the experiments carried out, the increases or decreases have different values for the vector. The **reward** is defined like distance-based in the manhattan distance between state to reach and actual state vectors, because manhattan distance will allow simple and efficient distance calculation between state vectors.

The ranges of minimum and maximum action values were also defined based on the experiments carried out in the research presented in Table 5.2. As an optimization policy,

the Bellman optimality equation was used for the learned value as in equation 5.5 cited by (Sutton and Barto 2017).

$$q(s,a) = R_{t+1} + \gamma * maxq(s',a')$$
(5.5)

Where q is the function of any state-action pair (s, a) at time t. R_{t+1} is the expected reward we get from taking action a in state s plus the maximum expected discounted return that can be achieved from any possible next state-action pair (s', a'). γ is the discount rate constant and maxqis the function of maximum value for action a' in state s' and with the value 0.99 in this research. It makes it possible to take into account the previous states, which will allow the algorithm to select the path with the least loss. At that time, for a new $q^{new}(s, a)$ is equal to a weighted sum of the last value and the learned value as it is shown in equation 5.6.

$$q^{new}(s,a) = (1-\alpha) * q^{old}(s,a) + \alpha [R_{t+1} + \gamma * maxq(s',a')]$$
(5.6)

Where α is a learning rate with value 0.1 in this research

In the next stages, it is necessary to define a geometry modeling simulation to validate the behavior of this technique.

5.3.1 Train weld bead geometry modeling

The training was carried out with the experimental data of the real GMAW process. The equation 5.6 is used an iterative process to calculate the *Q*-values, which are stored in a matrix table named *Q*-table with states x actions dimensions. *Q*-table guide to the best action for all possible states to obtain the objective state, like Figure 5.7 shows. It was defined the weld bead dimensions to goal, in this case (height = 2.20 mm, width = 7.73 mm, depth = -2.50 mm) The start action parameters over control variables, in this case (voltage = 20.0 v, wire-rate = 5.8 mm/s, welding velocity = 8.2 mm/s). The initial state for these parameters on stable situation are (height = 2.18 mm, width = 5.48 mm, depth = -2.95 mm) The number of episodes was 100 and the maximum steps by episode 100 to reach better control values. After obtaining *Q*-table can be found the actions to goal the objective weld bead dimensions by initial parameters.

Once the training has been carried out, it is possible to apply the Q-Table obtained to optimize the process, as shown in the Figure 5.8.



Figure 5.7: Algorithm of reinforcement learning train

Source: Produced by author



Figure 5.8: Algorithm of reinforcement learning model

E is the error allowed to the system. Source: Produced by author.

5.3.2 Drop Volume Methodology: simulation

The Figure 5.9 shows the simulation of the weld bead geometric parameters. It can be seen how after the 4th iteration the model reaches the three optimal values. The three geometric values obtain are the three minimums possible, permitted by the volume model, to goal the geometric objective.





The control error is: height = 0.68 mm, width = 0.76 mm and depth = 0.00 mm. Generating a margin of less than a millimeter for width and height, an approximation to the objective for depth. The Figure 5.10 shows control parameters. The variations of them in the firsts iterations do not expose sufficient change to make the system unstable.



5.3.3 Drop Frequency Methodology: simulation

The Figure 5.11 shows the geometric control process for drop frequency methodology. The geometric parameters obtained are not better than in the volume model, except the depth. The control error of this model is: height = 0.34 mm, width = 1.59 mm and depth = 0.32 mm, with only geometric parameters greater than one millimeter.

The Figure 5.12 shows the values obtained by the control parameters of the frequency model. Like the other model, this has small variations, so instability in the system should not be caused.



Figure 5.11: Drop Frequency Methodology: modeling geometric parameters



Figure 5.12: Drop Frequency Methodology: control parameters

5.4 CHAPTER CONSIDERATIONS

After analyzing the results in this chapter, it can conclude that the error margins obtained in the stages validate the methodologies and can be applied to a GMAW process with satisfactory results. The deep learning model obtained, proved that images can be classified, calculate unmelted wire length, the consumed wire volume, drop and short-circuit frequency in real-time for a GMAW process. This allows the use of these parameters in predictive and control models in real-time. The SVM algorithm is the one that offers the best results in the prediction of the GMAW process these two methodologies. The weal bead geometry modeling of GMAW process can be developed with the reinforcement learning technique due to the small margin of error that the simulation returns.

6 DESIGN AND DISCUSSIONS OF RESULTS

This chapter pretend shows the design proposed for the final system and the new technology used them. An analysis of the response times of this system will also be carried out. his chapter, the discussion of the results of this research work was developed.

6.1 DESIGN

The proposed design shows the equipment used for experiments in the GRACO laboratory of Brasilia University. It was proposed to add Jetson nano to process all modeling information, pass it to the computer, and communicate with the other equipment. It is shown in Figure 6.1. Jetson nano is an embedded system-on-module (SoM) and developer kit. Useful for deploying computer vision and deep learning, Jetson Nano runs Linux and provides 472 GFLOPS (floating-point operations, per second) compute performance with 5-10W of power consumption (eLinux 2020). The international cost is approximately 99 dollars.

This equipment reduces costs. It has excellent computational performance with low energy consumption. The average response time for frequency models and volume models are 0.25 ms and 18.25 ms respectively. In future developments, it may be possible to place the desktop pc functionalities in the Jetson nano. But for this, drivers of the welding source, the welding table, and the high-speed camera must be developed.

According to this, it necessary an additional network interface card in the computer. This must have the characteristics of maximum speed 1 Gps (gigabit per seconds), same as onboard. This would allow image transmission less than 0,5 ms. These cards have an approximate price of 7 dollars. The communication was developed through a web service, structured as shown Figure 6.2.

In the computer container, the welding supply driver and welding table driver are software packets developed in previous research. CxN packet is in charge gathers the image of the camera, sending it to the selected model, receiving the control parameters, and passing them to the welding supply and welding table. In jetson nano container, are both frequency-based models and volume-based models for geometry modeling of GMAW process. This action is repeated until the process is complete.

Based on the information placed, this entire process would have a minimum response time of 19.25 ms and an additional equipment cost less of 106 dollars. Therefore, the proposed design meets the objectives of low computational cost and low monetary cost.


Source: Produced by author

6.2 **DISCUSSIONS**

The objective of this research was to develop a weld bead geometry model of GMAW process by applying techniques of machine learning, deep learning, and reinforcement learning. The use of these computer science algorithms in welding processes is still new, although there are some applications. In the bibliographic review developed, the little presence of these practices in the welding processes was demonstrated, as shown in Figure 2.5 and Figure 2.6. In contrast to the advantages of these algorithms in various areas, solving problems similar to the main problems in welding processes and lined with new objectives of modern welding manufacturing presented by (Chen and Lv 2014). The potential of learning-based techniques was described because computational resources are becoming cheaper and welding information can be obtained more and more quality information. All these premises aligned with the industry 4.0, where a set of technologies that allow a fusion of physical and digital world to create more intelligent and dynamic systems, like (Brecher 2018) shown.

The articles analyzed where the presence of these algorithms is recognized, but sometimes they do not categorize them as machine learning techniques. The problem is that another series of similar techniques can be ignored and be optimal to generate a better result for a specific case. It



Source: Produced by author

was obvious in the little presence of model comparisons and infrequent applications of ANN and fuzzy techniques. In the paper's analysis was found a low application of deep learning techniques to process welding arc images. Reinforcement learning techniques have a low presence too. In this way, the novelty of applying these techniques in this investigative work is highlighted. In addition, the little interrelation between the nuclei authors and the few publications of these (Figure 3.3), demonstrate how novel the research are linking these areas.

Consequently, the arc parameters of unmelted wire length, molten wire, short-circuit frequency, droplet volume, and droplet frequency were selected. These parameters can be captured through image processing, which would allow testing these new deep learning techniques. To demonstrate these potentialities, a series of experiments was developed. Experiments 1 and 2 were planned to obtain acceptable variances in the welding parameters, which would allow obtaining a better predictive model. These variances were validated with the statistical analysis of the parameters shown in Table 5.2 and Table 5.3. This requirement is basic to be able to perform optimal modeling applying machine learning, like (Hernandez Orallo, Ramirez Quintana and Ferri Ramirez 2004, Witten and Frank 2005, Marbán, Mariscal and Segovia 2009) exposed.

Experiment 3 was focused on supporting the validation of the deep learning model, plus the approximate calculation of molten wire and droplet volume. The deep learning model was validated with an accuracy of 97,24% that defines the effectiveness of the model in performing a correct detection for both groups ("Short circuit and drop-detachment images" (SCDD) and "No short circuit and no drop-detachment images" (no SCDD)). Another metric that can be calculated with the data from the confusion matrix is Recall (95%). The recall measures the model's ability to detect positive samples. An important value for this research, since it performs the calculations of drop volume and molten wire volume from this detection.

In continuity with the development of the research, the Equation 5.4 was selected to calculate the volume. It is obtained as results of the equations presented by (Choi et al. 2001) and the conditions of droplet detachment detection. This equation makes it possible to calculate, with the

same formulation, the molten wire volume and the droplet volume. To allow the generalization of a model (deep learning + volume calculation) in the GMAW process for volume analysis. It is another novelty of this research since most of the volume calculations present in the literature have as a requirement to separate the analyzes by transfer modes. The equation is not exempt from calculation errors, as the author referenced. This error can be considered a systematic error that must usually be estimated by educated guesswork (Mortimer 2013). In turn, this error is assumed by machine learning models so that the permissible error of the model conceives the systematic error of the system. An estimate of the proper functioning of the process (deep learning + volume calculation) is the Figure 5.6. She clearly shows how the unmelted wire length (inverse of arc length) decreases and how the volume of molten wire decreases. It was the objective of experiment 3, and the process (deep learning + volume calculation) captured this function.

Once the arc parameters, the input parameters, and the geometric output parameters have been obtained, the tasks were started to obtain the predictive model. As the bibliographic analysis showed, the predictive model is one of the three learning models presented, the most applied in welding processes. But as a consequence, a generalized two models of the GMAW process were obtained for all transfer modes. A comparison was also made between the two predictive models developed. The Table 6.1 shows a summary of it.

	Drop Volume Methodology	Drop Frequency Methodology
different parameters	unmelted wire length, melted wire	short-circuit frequency, droplet
	volume, droplet volume	detachment frequency
Predictive-model	0.27	0.33
(SVM-RBF) RSME		
Control error	height = 0.68 mm, width = 0.76	height = 0.34 mm, width = 1.59
	mm, depth = 0.00 mm	mm, depth = 0.32 mm
Performance time	130 ms	11 ms
(Laptop)		
Performance time (Jetson	18.25 ms	0.25 ms
nano)		

Table 6.1: Comparison of the two methodologies

As already expressed, this small difference in the RSME means that any of the two predictive models can be used. Always taking into account the requirements of the system in which it will be applied. Another relevance of this predictive model is that it was proposed to predict three parameters of the geometry of the weld bead at the same time.

The processes, deep learning + (volume or frequency calculation) + predictive model, make up the necessary environment for the reinforcement learning process. As result, the drop volume methodology offers fewer error margins for each geometric parameters, like Table 6.1 shows. But doing the comparison based on response time, the drop frequency methodology has 0.011 seconds on average. Better than the drop volume methodology that has 0.13 seconds. These tests were made on a laptop with RAM: 8 GB, processors: 1.70 GHz x 4, without a dedicated graphics card. Another run test was performed on a Jetson nano. The result was the one shown in the Table 6.1. These increased computational resources improve the run-time of the drop volume methodology.

It would make it more feasible in a production environment but at the same time is less cheaper than the drop frequency methodology. Conversely, the drop frequency methodology would be cheaper by the industry but would have a margin of error that may not be accepted. Another important analysis in drop volume methodology is the drop calculation. Despite being based on formulations of a reference article, this article could introduce another possible errors due to the complexity of the process that occurs in the welding arc.

Others intelligent control techniques, like ANN-fuzzy, has the problem that specialist has to know and define the rules associated to process, which is few probable based on all geometric parameters control and the complex relation between them. (Cruz, Torres and Alfaro 2015) defined 12 fuzzy functions and 7 fuzzy rules for only width control. These fuzzy rules complexities will rise with other control parameters. Q-learning has not this problem how was exposed in this research. The ANN used on intelligent control increases the computational cost compared to other machine learning techniques that can offer the better or same result. The results obtained allow the use of these techniques with good computational performance in terms of training and execution.

7 CONCLUSION

- A bibliographic analysis was developed that showed the new requirements for an intelligent welding process, and the potentialities of machine learning techniques to solve many of these problems were demonstrated. Although this research focused on the GMAW process, this research opens the door to these techniques in other welding processes.
- An efficient resnet model was developed that allows the detection of droplet detachment and the occurrence of short circuits in real-time for a GMAW process. The use of background subtraction technique allowed the visualization of the detachment pattern.
- Functionality was developed with a deep learning model and a physical analysis of the GMAW process, which make it possible to calculate a droplet volume and a molten wire in real-time. Despite the possible systematic error inserted by the deep learning model and the physical formulation used, this analysis can be the beginning of several studies jointly applying physical analyzes and deep learning techniques in the welding arc.
- Two models were developed with the Support Vector Machine algorithm with Radial Basis Functionality kernel, giving a low root-mean-square error, showing the possibility of predicting a GMAW process for all transfer modes.
- The two proposed methodologies are analyzed and compared. Its advantages and disadvantages were shown according to its possible application in research and industrial projects. The times obtained using a Jetson nano facilitate the application of either of the two methodologies. In addition, the use of jetson nano potentiates the application of the internet of things (IoT) associated with welding processes.
- A system design proposal was made based on the equipment used in the research, where any of the two proposed methodologies can be applied. An analysis of the computational cost of the entire system and the monetary cost was developed. In conclusion, the feasibility of developing the system and the potential to carry out new research were demonstrated.

8 FUTURE WORKS

- Perform laboratory tests of these methodologies in a GMAW process. It would also lead to the development the proposed system and develop all control software on jetson nano.
- Applying these deep learning techniques and use more modern video sensors, a more in-depth analysis of the arc can be carried out, deepening the studies of molten material, drop analysis, and material losses.
- Extend the techniques analysis of arc welding to other welding processes, like Gas tungsten arc welding (GTAW), Plasma arc welding (PAW), Shielded metal arc welding (SMAW), Submerged arc welding (SAW).
- To develop a research in which all visual events of the welding arc are correlated with the characteristic sound. It would later allow the development of models based on the arc sound.
- To develop other models with other parameters and make comparisons between them to find the most efficient one. Approach the study of the rest of the parameters and their interrelation by applying machine learning techniques.
- To develop other models of reinforcement learning. Compare them and the models proposed in this research.
- To investigate real-time weld bead geometry sensors to be incorporated into intelligent control research. It would simplify the applications of the models and could provide better efficiency.
- To develop a study on the analysis and application of machine learning techniques in orbital welding.
- To develop an investigation of potential internet of things (IoT) and their technologies that can be applied to welding processes.
- To Develop an investigation applying the reinforcement learning technique with an online training.

BIBLIOGRAPHIC REFERENCES

Agarwal 2013 AGARWAL, S. Data mining: Data mining concepts and techniques. In: . Piscataway: The Institute of Electrical and Electronics Engineers, Inc. (IEEE), 2013. p. 203–207. Copyright - Copyright The Institute of Electrical and Electronics Engineers, Inc. (IEEE) Dec 2013; Última actualización - 2015-05-29. Available from Internet: https://search.proquest.com/docview/1684095527?accountid=26646>.

Alpaydin 2010 ALPAYDIN, E. *Introduction to machine learning*. USA: Massachusetts Institute of Technology, 2010. ISBN 978-0-262-01243-0.

Alvarez Bestard and Absi Alfaro 2018 Alvarez Bestard, G.; Absi Alfaro, S. C. Measurement and estimation of the weld bead geometry in arc welding processes: the last 50 years of development. *Journal of the Brazilian Society of Mechanical Sciences and Engineering*, Springer Berlin Heidelberg, v. 40, n. 9, p. 444, sep 2018. ISSN 1678-5878. Available from Internet: http://link.springer.com/10.1007/s40430-018-1359-2.

Analytics 2018 ANALYTICS, C. *Web of Science*. 2018. Available from Internet: http://webofknowledge.com>.

Arel, Rose and Karnowski 2010 AREL, I.; ROSE, D. C.; KARNOWSKI, T. P. Research frontier: Deep machine learning–a new frontier in artificial intelligence research. *Comp. Intell. Mag.*, IEEE Press, Piscataway, NJ, USA, v. 5, n. 4, p. 13–18, nov. 2010. ISSN 1556-603X. Available from Internet: http://dx.doi.org/10.1109/MCI.2010.938364>.

ASM 1994 ASM. Gas-metal arc welding. Welding Handbook. 2. ed. [S.l.: s.n.], 1994. 569–581 p.

Aviles-Viñas, Rios-Cabrera and Lopez-Juarez 2016 AVILES-VIÑAS, J. F.; RIOS-CABRERA, R.; LOPEZ-JUAREZ, I. On-line learning of welding bead geometry in industrial robots. *International Journal of Advanced Manufacturing Technology*, v. 83, n. 1-4, p. 217–231, 2016. ISSN 14333015.

Azadi Moghaddam, Golmezergi and Kolahan 2016 Azadi Moghaddam, M.; GOLMEZERGI, R.; KOLAHAN, F. Multi-variable measurements and optimization of GMAW parameters for API-X42 steel alloy using a hybrid BPNN–PSO approach. *Measurement*, Elsevier, v. 92, p. 279–287, oct 2016. ISSN 0263-2241. Available from Internet: https://www.sciencedirect.com/science/article/pii/S0263224116302056>.

Bagavathiappan et al. 2013 BAGAVATHIAPPAN, S.; LAHIRI, B. B.; SARAVANAN, T.; PHILIP, J.; JAYAKUMAR, T. Infrared thermography for condition monitoring - A review. *Infrared physics & technology*, ELSEVIER SCIENCE BV, PO BOX 211, 1000 AE AMSTERDAM, NETHERLANDS, v. 60, p. 35–55, sep 2013. ISSN 1350-4495.

Bai and Lubecki 2016 BAI, F.; LUBECKI, T. M. Robotic Arc Welding with On-line Process Monitoring Based on the LMM Analysis of the Welding Process Stability. *2016 IEEE International Conference on Advanced Intelligent Mechatronics (AIM)*, IEEE, p. 566–571, jul 2016. Available from Internet: http://ieeexplore.ieee.org/document/7576828/>.

Balsamo et al. 2000 BALSAMO, P.; VILARINHO, L.; VILELA, M.; SCOTTI, A. Development of an experimental technique for studying metal transfer in welding: Synchronised shadowgraphy. *Int. J. Join. Mater.*, v. 12, 06 2000.

Baraka, Panoutsos and Cater 2015 BARAKA, A.; PANOUTSOS, G.; CATER, S. A real-time quality monitoring framework for steel friction stir welding using computational intelligence. *Journal of*

Manufacturing Processes, The Society of Manufacturing Engineers, v. 20, p. 137–148, 2015. ISSN 15266125. Available from Internet: http://dx.doi.org/10.1016/j.jmapro.2015.09.001>.

Bell 2015 BELL, J. *Machine Learning: Hands-On for Developers and Technical Professionals.* Indianapolis, IN, USA: John Wiley & Sons, Inc., 2015. ISBN 978-1-118-88906-0.

Bestard et al. 2018 BESTARD, G.; SAMPAIO, R.; VARGAS, J.; ALFARO, S.; BESTARD, G. A.; SAMPAIO, R. C.; VARGAS, J. A. R.; ALFARO, S. C. A. Sensor Fusion to Estimate the Depth and Width of the Weld Bead in Real Time in GMAW Processes. *Sensors*, Multidisciplinary Digital Publishing Institute, v. 18, n. 4, p. 962, mar 2018. ISSN 1424-8220. Available from Internet: http://www.mdpi.com/1424-8220/18/4/962>.

Bestard 2017 BESTARD, G. A. Sensor fusion and embedded devices to estimate and control the depth and width of the weld bead in real time. Ph.D. thesis. 151 p. Tese (Doutorado), nov 2017. Available from Internet: http://repositorio.unb.br/handle/10482/31429>.

Bingul and Cook 2006 BINGUL, Z.; COOK, G. A real-time prediction model of electrode extension for GMAW. *IEEE/ASME Transactions on Mechatronics*, v. 11, n. 1, p. 47–54, feb 2006. ISSN 1083-4435. Available from Internet: http://ieeexplore.ieee.org/document/1597175/.

Boutaghane et al. 2011 BOUTAGHANE, A.; BOUHADEF, K.; VALENSI, F.; PELLERIN, S.; BENKEDDA, Y. Theoretical model and experimental investigation of current density boundary condition for welding arc study. *European physical journal-applied physics*, CAMBRIDGE UNIV PRESS, 32 AVENUE OF THE AMERICAS, NEW YORK, NY 10013-2473 USA, v. 54, n. 1, 2011. ISSN 1286-0042.

Bouwmans, Aybat and ZAHZAH 2016 BOUWMANS, T.; AYBAT, N.; ZAHZAH, E.-h. *Handbook on* "*Robust Low-Rank and Sparse Matrix Decomposition: Applications in Image and Video Processing*". [S.l.: s.n.], 2016.

Bouwmans et al. 2014 BOUWMANS, T.; PORIKLI, F.; HöFERLIN, B.; VACAVANT, A. Handbook on "Background Modeling and Foreground Detection for Video Surveillance". [S.I.: s.n.], 2014.

Brecher 2018 BRECHER, C. SMART DATA FOR BETTER DECISIONS HOW DIGITIZATION IS CHANGING PRODUCTION TECHNOLOGY. 2018. Available from Internet: https://www.fronius.com/en/welding-technology/info-centre/magazine/2018/smart-data.

Breheny 2016 BREHENY, P. Ridge Regression. In: . [S.l.: s.n.], 2016. p. 1–22.

Cary and Helzer 2005 CARY, H. B.; HELZER, S. C. *Modern welding technology*. Prentice Hall, 2005. 715 p. ISBN 0131130293. Available from Internet: https://www.abebooks.com/9780132418034/ Modern-Welding-Technology-4th-Edition-0132418037/plp>.

Casalino 2018 CASALINO, G. [INVITED] Computational intelligence for smart laser materials processing. *Optics & Laser Technology*, v. 100, p. 165–175, mar 2018. ISSN 00303992. Available from Internet: https://linkinghub.elsevier.com/retrieve/pii/S0030399217303286>.

Cayo 2013 CAYO, E. H. Modelagem e controle da soldagem orbital através do processo GMAW-S. jan 2013. Available from Internet: http://repositorio.unb.br/handle/10482/1444.

Chambers, Doig and Stokes-Rees 2017 CHAMBERS, M.; DOIG, C.; STOKES-REES, I. *Breaking Data Science Open*. [S.1.: s.n.], 2017. 1–88 p. ISBN 978-1-491-97299-1.

Chan and Vasconcelos 2012 CHAN, A. B.; VASCONCELOS, N. Counting People With Low-Level Features and Bayesian Regression. *IEEE Transactions on Image Processing*, v. 21, n. 4, p. 2160–2177, apr 2012. ISSN 1057-7149. Available from Internet: http://ieeexplore.ieee.org/document/6054049/>.

Chapman et al. 2000 CHAPMAN, P.; CLINTON, J.; KERBER, R.; KHABAZA, T.; REINARTZ, T.; SHEARER, C.; WIRTH, R. CRISP-DM 1.0. *The CRISP-DM consortium*, ago 2000.

Chen and Chen 2010 CHEN, B.; CHEN, S. Multi-sensor information fusion in pulsed GTAW based on fuzzy measure and fuzzy integral. *Assembly automation*, EMERALD GROUP PUBLISHING LIMITED, HOWARD HOUSE, WAGON LANE, BINGLEY BD16 1WA, W YORKSHIRE, ENGLAND, v. 30, n. 3, p. 276–285, 2010. ISSN 0144-5154.

Chen and Feng 2014 CHEN, B.; FENG, J. Modeling of underwater wet welding process based on visual and arc sensor. *Industrial robot-an international journal*, EMERALD GROUP PUBLISHING LIMITED, HOWARD HOUSE, WAGON LANE, BINGLEY BD16 1WA, W YORKSHIRE, ENGLAND, v. 41, n. 3, p. 311–317, 2014. ISSN 0143-991X.

Chen, Wang and Chen 2010 CHEN, B.; WANG, J.; CHEN, S. Prediction of pulsed GTAW penetration status based on BP neural network and D-S evidence theory information fusion. *International journal of advanced manufacturing technology*, SPRINGER LONDON LTD, 236 GRAYS INN RD, 6TH FLOOR, LONDON WC1X 8HL, ENGLAND, v. 48, n. 1-4, p. 83–94, 2010. ISSN 0268-3768.

Chen and Farson 2010 CHEN, J. Z.; FARSON, D. F. Hybrid Welds Coaxial Vision Monitoring of LBW/GMAW Hybrid Welding Process. *Materials evaluation*, AMER SOC NONDESTRUCTIVE TEST, 1711 ARLINGATE LANE PO BOX 28518, COLUMBUS, OH 43228-0518 USA, v. 68, n. 12, p. 1318–1328, 2010. ISSN 0025-5327.

Chen et al. 2000 CHEN, S. B.; LOU, Y. J.; WU, L.; ZHAO, D. B. Intelligent methodology for sensing, modeling and control of pulsed GTAW: Part I - Bead-on-plate welding. *Welding journal*, v. 79, n. 6, p. 151s–163s, 2000. ISSN 00432296.

Chen and Lv 2014 CHEN, S. B.; LV, N. Research evolution on intelligentized technologies for arc welding process. *Journal of Manufacturing Processes*, v. 16, n. 1, p. 109–122, 2014. ISSN 15266125.

Chen, Wang and Ma 2010 CHEN, S. B.; WANG, W. Y.; MA, H. B. Intelligent control of arc welding dynamics during robotic welding process. In: Chandra, T and Wanderka, N and Reimers, W and Ionescu, M (Ed.). *Thermec 2009, PTS 1-4*. LAUBLSRUTISTR 24, CH-8717 STAFA-ZURICH, SWITZERLAND: TRANS TECH PUBLICATIONS LTD, 2010. (Materials Science Forum, 1-4), p. 3751–3756. ISSN 0255-5476.

Chi et al. 2019 Chi, R.; Hou, Z.; Jin, S.; Huang, B. An improved data-driven point-to-point ilc using additional on-line control inputs with experimental verification. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, v. 49, n. 4, p. 687–696, April 2019. ISSN 2168-2216.

Chincoli and Liotta 2018 CHINCOLI, M.; LIOTTA, A. Self-Learning Power Control in Wireless Sensor Networks. *Sensors*, v. 18, n. 2, p. 375, jan 2018. ISSN 1424-8220. Available from Internet: http://www.mdpi.com/1424-8220/18/2/375>.

Chiumenti et al. 2013 CHIUMENTI, M.; CERVERA, M.; Agelet de Saracibar, C.; DIALAMI, N. Numerical modeling of friction stir welding processes. *Computer Methods in Applied Mechanics and Engineering*, North-Holland, v. 254, p. 353–369, feb 2013. ISSN 0045-7825. Available from Internet: https://www.sciencedirect.com/science/article/pii/S0045782512002952>.

Choi et al. 2001 CHOI, J. H.; CHOI, J. H.; LEE, J. Y.; LEE, J. Y.; YOO, C. D.; YOO, C. D. Simulation of Dynamic Behavior in a GMAW System. *Mechanical Engineering*, n. October, p. 239–245, 2001. ISSN 00432296.

Chong, Ramakrishna and Singh 2018 CHONG, L.; RAMAKRISHNA, S.; SINGH, S. A review of digital manufacturing-based hybrid additive manufacturing processes. *The International Journal of Advanced Manufacturing Technology*, v. 95, n. 5-8, p. 2281–2300, mar 2018. ISSN 0268-3768. Available from Internet: http://link.springer.com/10.1007/s00170-017-1345-3.

Chun, Park and Choi 2018 CHUN, T. Y.; PARK, J. B.; CHOI, Y. H. Reinforcement Q-learning based on Multirate Generalized Policy Iteration and Its Application to a 2-DOF Helicopter. *International Journal of Control, Automation and Systems*, v. 16, n. 1, p. 377–386, feb 2018. ISSN 20054092. Available from Internet: http://link.springer.com/10.1007/s12555-017-0172-5.

Coniglio et al. 2016 CONIGLIO, N.; MATHIEU, A.; AUBRETON, O.; STOLZ, C. Weld pool surface temperature measurement from polarization state of thermal emission. *Quantitative infrared thermography journal*, TAYLOR & FRANCIS LTD, 4 PARK SQUARE, MILTON PARK, ABINGDON OX14 4RN, OXON, ENGLAND, v. 13, n. 1, p. 83–93, 2016. ISSN 1768-6733.

Cortes and Vapnik 1995 CORTES, C.; VAPNIK, V. Support-vector networks. *Machine Learning*, Kluwer Academic Publishers, v. 20, n. 3, p. 273–297, sep 1995. ISSN 0885-6125. Available from Internet: http://link.springer.com/10.1007/BF00994018>.

Cruz, Torres and Alfaro 2015 CRUZ, J. G.; TORRES, E. M.; ALFARO, S. C. A. A methodology for modeling and control of weld bead width in the GMAW process. *Journal of the brazilian society of mechanical sciences and engineering*, SPRINGER HEIDELBERG, TIERGARTENSTRASSE 17, D-69121 HEIDELBERG, GERMANY, v. 37, n. 5, p. 1529–1541, sep 2015. ISSN 1678-5878.

Currin et al. 1991 CURRIN, C.; MITCHELL, T.; MORRIS, M.; YLVISAKER, D. Bayesian Prediction of Deterministic Functions, with Applications to the Design and Analysis of Computer Experiments. *Journal of the American Statistical Association*, v. 86, n. 416, p. 953–963, dec 1991. ISSN 0162-1459. Available from Internet: http://www.tandfonline.com/doi/abs/10.1080/01621459.1991.10475138>.

Deeplizard 2019 DEEPLIZARD. *Reinforcement Learning - Introducing Goal Oriented Intelligence*. 2019. Available from Internet: http://deeplizard.com/learn/video/my207WNoeyA.

Díaz Franco 2008 Díaz Franco, F. U. d. B. *Monitorização e localização de defeitos na soldagem TIG através do sensoriamento infravermelho*. Tese (Doutorado), 2008.

Dishion et al. 2004 DISHION, T. J.; NELSON, S. E.; WINTER, C. E.; BULLOCK, B. M. Adolescent friendship as a dynamic system: Entropy and deviance in the etiology and course of male antisocial behavior. *Journal of abnormal child psychology*, v. 32, n. 6, p. 651–63, 12 2004. Copyright - Copyright (c) 2004 Springer Science+Business Media, Inc; Última actualización - 2014-07-26; CODEN - JABCAA. Available from Internet: https://search.proquest.com/docview/204989070?accountid=26646>.

Dong et al. 2016 DONG, H.; CONG, M.; LIU, Y.; ZHANG, Y.; CHEN, H. Predicting characteristic performance for arc welding process. In: 2016 IEEE international conference on cyber technology in automation, control and intelligent systems (CYBER). IEEE, 2016. p. 7–12. ISBN 978-1-5090-2733-0. Available from Internet: http://ieeexplore.ieee.org/document/7574786/>.

Elgendy 2020 ELGENDY, M. *Deep Learning for Vision Systems*. 2020. Available from Internet: https://livebook.manning.com/book/grokking-deep-learning-for-computer-vision/chapter-1/45>.

eLinux 2020 ELINUX. Jetson Nano. 2020. Available from Internet: ">https://elinux.org/Jetson{_N">https://e

Escribano-García et al. 2014 ESCRIBANO-GARCÍA, R.; LOSTADO-LORZA, R.; FERNÁNDEZ-MARTÍNEZ, R.; VILLANUEVA-ROLDÁN, P.; Mac Donald, B. J. Improvement in Manufacturing Welded Products Through Multiple Response Surface Methodology And Data Mining Techniques. *Advances in Intelligent Systems and Computing*, v. 299, p. 301–310, 2014. Available from Internet: <http://link.springer.com/10.1007/978-3-319-07995-0{_}30https://www.scopus.com/inward/record.uri?eid=2-s2.0-84927711846{&}partnerID=40{&}md5=93041e2e73c7e160001be7d>.

EuroTech 2019 EUROTECH. *Gas Metal Arch Welding (GMAW) Process*. 2019. Available from Internet: http://eurotechworld.net/trainings/welder-training/gas-metal-arch-welding-gmaw-process/.

Fayyad et al. 1996 FAYYAD, U. M.; PIATETSKY-SHAPIRO, G.; SMYTH, P.; UTHURUSAMY, R. *Advances in knowledge discovery and data mining*. [S.l.]: AAAI press Menlo Park, 1996. v. 21.

Fearn 2013 FEARN, T. Ridge Regression. *NIR news*, v. 24, n. 3, p. 18–19, 2013. ISSN 0960-3360. Available from Internet: http://journals.sagepub.com/doi/10.1255/nirn.1365>.

Fei Gao and Qinglan Chen 2015 Fei Gao; Qinglan Chen. Study on arc welding robot weld seam touch sensing location method for structural parts of hull. In: *2015 International Conference on Control, Automation and Information Sciences (ICCAIS)*. IEEE, 2015. p. 42–46. ISBN 978-1-4799-9892-0. Available from Internet: ">http://ieeexplore.ieee.org/document/7338704/>.

Felzenszwalb et al. 2010 FELZENSZWALB, P. F.; GIRSHICK, R. B.; MCALLESTER, D.; RAMANAN, D. Object detection with discriminatively trained part-based models. *IEEE Trans. Pattern Anal. Mach. Intell.*, IEEE Computer Society, Washington, DC, USA, v. 32, n. 9, p. 1627–1645, set. 2010. ISSN 0162-8828. Available from Internet: http://dx.doi.org/10.1109/TPAMI.2009.167>.

Feng et al. 2012 FENG, S.; LIN, G.; MA, B.; HU, S. A novel measurement and qualification method of GMAW welding fault based on digital signals. In: Chen, WZ and Xu, XP and Dai, PQ and Chen, YL (Ed.). *Advanced manufacturing technology, pts 1-4.* LAUBLSRUTISTR 24, CH-8717 STAFA-ZURICH, SWITZERLAND: TRANS TECH PUBLICATIONS LTD, 2012. (Advanced Materials Research, v. 472-475), p. 1201–1205. ISBN 978-3-03785-370-2. ISSN 1022-6680.

Fidali and Jamrozik 2013 FIDALI, M.; JAMROZIK, W. Diagnostic method of welding process based on fused infrared and vision images. *Infrared physics & technology*, ELSEVIER SCIENCE BV, PO BOX 211, 1000 AE AMSTERDAM, NETHERLANDS, v. 61, p. 241–253, nov 2013. ISSN 1350-4495.

Friedman, Hastie and Tibshirani 2010 FRIEDMAN, J.; HASTIE, T.; TIBSHIRANI, R. Regularization Paths for Generalized Linear Models via Coordinate Descent. v. 33, n. 1, 2010. ISSN 0014-4886.

Fronius 2012 FRONIUS. *CMT 4000 ADVANCED*, *TIME 5000 DIGITAL*, *TRANSPULS SYNERGIC 2700/3200/4000/5000*. [S.1.: s.n.], 2012.

Gao et al. 2011 GAO, X.; DING, D.; BAI, T.; KATAYAMA, S. Weld-pool image centroid algorithm for seam-tracking vision model in arc-welding process. *IET Image processing*, INST ENGINEERING TECHNOLOGY-IET, MICHAEL FARADAY HOUSE SIX HILLS WAY STEVENAGE, HERTFORD SG1 2AY, ENGLAND, v. 5, n. 5, p. 410–419, 2011. ISSN 1751-9659.

Gill 2018 GILL, J. K. *Automatic Log Analysis using Deep Learning and AI for Microservices*. 2018. Available from Internet: https://www.xenonstack.com/blog/log-analytics-deep-machine-learning/.

Giron Cruz 2014 Giron Cruz, J. A. *Uma metodologia para modelagem e controle da altura do reforço e da largura do cordão de solda no processo gmaw*. Tese (Doutorado), 2014.

Günther et al. 2014 GÜNTHER, J.; PILARSKI, P. M.; HELFRICH, G.; SHEN, H.; DIEPOLD, K. First Steps Towards an Intelligent Laser Welding Architecture Using Deep Neural Networks and Reinforcement Learning. *Procedia Technology*, v. 15, p. 474–483, 2014. ISSN 22120173. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S2212017314001224>.

Günther et al. 2016 GÜNTHER, J.; PILARSKI, P. M.; HELFRICH, G.; SHEN, H.; DIEPOLD, K. Intelligent laser welding through representation, prediction, and control learning: An architecture with deep neural networks and reinforcement learning. *Mechatronics*, v. 34, p. 1–11, mar 2016. ISSN 09574158. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0957415815001555>.

Haffner et al. 2017 HAFFNER, O.; KUCERA, E.; KOZAK, S.; STARK, E. Proposal of system for automatic weld evaluation. In: *2017 21st International Conference on Process Control* (*PC*). IEEE, 2017. p. 440–445. ISBN 978-1-5386-4011-1. Available from Internet: http://ieeexplore.ieee.org/document/7976254/>.

Hailin et al. 2012 HAILIN, H.; JING, L.; FANG, L.; WEI, Z.; HEQIANG, P. Neural-fuzzy variable gap control method for GMAW pipe-line welding with CCD camera. In: Zhao, H (Ed.). *Mechanical and electronics engineering III, pts 1-5.* LAUBLSRUTISTR 24, CH-8717 STAFA-ZURICH, SWITZERLAND: TRANS TECH PUBLICATIONS LTD, 2012. (Applied Mechanics and Materials, 1-5), p. 2358–2363. ISBN 978-3-03785-286-6. ISSN 1660-9336.

He et al. 2015 HE, K.; ZHANG, X.; REN, S.; SUN, J. Deep Residual Learning for Image Recognition. dec 2015. Available from Internet: http://arxiv.org/abs/1512.03385>.

Hernandez Orallo, Ramirez Quintana and Ferri Ramirez 2004 Hernandez Orallo, J.; Ramirez Quintana, M. J.; Ferri Ramirez, C. *Introduccion a la mineria de datos*. [S.l.]: Pearson Prentice Hall, 2004. ISBN 8420540919.

Hirji 1999 HIRJI, K. K. Discovering data mining: From concept to implementation. *SIGKDD Explor*. *Newsl.*, ACM, New York, NY, USA, v. 1, n. 1, p. 44–45, jun. 1999. ISSN 1931-0145. Available from Internet: http://doi.acm.org/10.1145/846170.846181.

Horvat et al. 2011 HORVAT, J.; PREZELJ, J.; POLAJNAR, I.; ČUDINA, M. Monitoring Gas Metal Arc Welding Process by Using Audible Sound Signal. *Strojniški vestnik – Journal of Mechanical Engineering*, v. 2011, n. 03, p. 267–278, mar 2011. ISSN 00392480. Available from Internet: .

Hou et al. 2018 HOU, W.; WEI, Y.; GUO, J.; JIN, Y.; ZHU, C. Automatic Detection of Welding Defects using Deep Neural Network. *Journal of Physics: Conference Series*, v. 933, p. 012006, jan 2018. ISSN 1742-6588. Available from Internet: http://stacks.iop.org/1742-6596/933/i=1/a=012006?key=crossref. b46613e0249463ae0a8765d3f39d1edf>.

Hu, Huang and Zeng 2017 HU, P.; HUANG, J.; ZENG, M. Application of fuzzy control method in gas metal arc welding. *The International Journal of Advanced Manufacturing Technology*, The International Journal of Advanced Manufacturing Technology, v. 92, n. 5-8, p. 1769–1775, 2017. ISSN 0268-3768. Available from Internet: http://link.springer.com/10.1007/s00170-017-0245-x.

Huang et al. 2017 HUANG, Y.; WU, D.; LV, N.; CHEN, H.; CHEN, S. Investigation of porosity in pulsed GTAW of aluminum alloys based on spectral and X-ray image analyses. *Journal of Materials Processing Technology*, v. 243, p. 365–373, may 2017. ISSN 09240136. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0924013616304691>.

Huang et al. 2019 Huang, Z.; Xu, X.; He, H.; Tan, J.; Sun, Z. Parameterized batch reinforcement learning for longitudinal control of autonomous land vehicles. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, v. 49, n. 4, p. 730–741, April 2019. ISSN 2168-2216.

INRIA INRIA. *scikit-learn: machine learning in Python* — *scikit-learn* 0.20.0 *documentation*. Available from Internet: ">https://scikit-learn.org/stable/.

Jiang, Zhang and Wang 2017 JIANG, C.; ZHANG, F.; WANG, Z. Image Processing of Aluminum Alloy Weld Pool for Robotic VPPAW Based on Visual Sensing. *IEEE Access*, v. 5, p. 21567–21573, 2017. ISSN 2169-3536. Available from Internet: http://ieeexplore.ieee.org/document/8064625/.

Kalaichelvi, Karthikeyan and Sivakumar 2013 KALAICHELVI, V.; KARTHIKEYAN, R.; SIVAKUMAR, D. Analysis of gas metal arc welding process using GA tuned fuzzy rule based system. *Journal of Intelligent & Fuzzy Systems*, IOS Press, v. 25, n. 2, p. 429–440, jan 2013. ISSN 1064-1246. Available from Internet: https://content.iospress.com/articles/journal-of-intelligent-and-fuzzy-systems/ ifs650>.

Keshmiri et al. 2015 KESHMIRI, S.; ZHENG, X.; FENG, L. W.; PANG, C. K.; CHEW, C. M. Application of deep neural network in estimation of the weld bead parameters. In: *IEEE International Conference on Intelligent Robots and Systems*. [s.n.], 2015. v. 2015-Decem, p. 3518–3523. ISBN 9781479999941. ISSN 21530866. Available from Internet: http://arxiv.org/abs/1502.04187>.

Kim, Park and Sohmshetty 2017 KIM, K. Y.; PARK, J.; SOHMSHETTY, R. Prediction measurement with mean acceptable error for proper inconsistency in noisy weldability prediction data. *Robotics and Computer-Integrated Manufacturing*, v. 43, p. 18–29, feb 2017. ISSN 07365845. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0736584516300187>.

Kingma and Ba 2015 KINGMA, D. P.; BA, J. L. Adam: A method for stochastic optimization. In: *International Conference on Learning Representations*. [S.l.: s.n.], 2015. p. 1–15.

Krizhevsky, Sutskever and Hinton 2017 KRIZHEVSKY, A.; SUTSKEVER, I.; HINTON, G. E. Imagenet classification with deep convolutional neural networks. *Commun. ACM*, Association for Computing Machinery, New York, NY, USA, v. 60, n. 6, p. 84–90, maio 2017. ISSN 0001-0782. Available from Internet: https://doi.org/10.1145/3065386>.

Kumar et al. 2014 KUMAR, G. S.; NATARAJAN, U.; VEERARAJAN, T.; ANANTHAN, S. S. Quality Level Assessment for Imperfections in GMAW. *Welding journal*, AMER WELDING SOC, 550 N W LEJEUNE RD, MIAMI, FL 33126 USA, v. 93, n. 3, p. 85S–97S, mar 2014. ISSN 0043-2296.

Kumar, Anand and Srivastava 2014 KUMAR, J.; ANAND, R. S.; SRIVASTAVA, S. P. Flaws classification using ANN for radiographic weld images. In: *2014 International Conference on Signal Processing and Integrated Networks (SPIN)*. IEEE, 2014. p. 145–150. ISBN 978-1-4799-2866-8. Available from Internet: http://ieeexplore.ieee.org/document/6776938/.

Kumar et al. 2016 KUMAR, V.; ALBERT, S. K.; CHANDRASEKHAR, N.; JAYAPANDIAN, J.; VENKATESAN, M. V. Performance analysis of arc welding parameters using self organizing maps and probability density distributions. In: 2016 IEEE First International Conference on Control, Measurement and Instrumentation (CMI). IEEE, 2016. p. 196–200. ISBN 978-1-4799-1769-3. Available from Internet: http://ieeexplore.ieee.org/document/7413738/.

Larose 2014 LAROSE, D. T. *Discovering knowledge in data: an introduction to data mining*. [S.l.]: John Wiley & Sons, 2014.

LeCun, Bengio and Hinton 2015 LECUN, Y.; BENGIO, Y.; HINTON, G. Deep learning. *Nature*, Nature Publishing Group, v. 521, n. 7553, p. 436–444, may 2015. ISSN 0028-0836. Available from Internet: http://www.nature.com/doifinder/10.1038/nature14539>.

Lecun et al. 1998 Lecun, Y.; Bottou, L.; Bengio, Y.; Haffner, P. Gradient-based learning applied to document recognition. *Proceedings of the IEEE*, v. 86, n. 11, p. 2278–2324, 1998.

Li et al. 2018 LI, J.; CHAI, T.; LEWIS, F. L.; FAN, J.; DING, Z.; DING, J. Off-Policy Q-Learning: Set-Point Design for Optimizing Dual-Rate Rougher Flotation Operational Processes. *IEEE Transactions on Industrial Electronics*, v. 65, n. 5, p. 4092–4102, may 2018. ISSN 02780046. Available from Internet: http://ieeexplore.ieee.org/document/8061021/>.

Li et al. 2014 LI, W.; GAO, K.; WU, J.; HU, T.; WANG, J. SVM-based information fusion for weld deviation extraction and weld groove state identification in rotating arc narrow gap MAG welding. *International journal of advanced manufacturing technology*, SPRINGER LONDON LTD, 236 GRAYS INN RD, 6TH FLOOR, LONDON WC1X 8HL, ENGLAND, v. 74, n. 9-12, p. 1355–1364, oct 2014. ISSN 0268-3768.

Li and Gao 2014 LI, Z.; GAO, X. Study on Regression Model of Measuring Weld Position. In: Choi, SB and Yarlagadda, P and AbdullahAlWadud, M (Ed.). *Sensors, mechatronics and automation*. LAUBLSRUTISTR 24, CH-8717 STAFA-ZURICH, SWITZERLAND: TRANS TECH PUBLICATIONS LTD, 2014. (Applied Mechanics and Materials, v. 511-512), p. 514–517. ISBN 978-3-03835-017-0. ISSN 1660-9336.

Liu et al. 2012 LIU, T.-k.; YU, J.-l.; CHEN, C.-l.; WEI, P.-s. Information theoretic perspective on coastal water-quality monitoring and management near an offshore industrial park. *Environmental monitoring and assessment*, v. 184, n. 8, p. 4725–35, 08 2012. Copyright - Springer Science+Business Media B.V. 2012; Características del documento - ; Equations; Tables; Última actualización - 2014-08-09; SubjectsTermNotLitGenreText - Taiwan. Available from Internet: https://search.proquest.com/docview/1023189660?accountid=26646>.

Liu, Huang and Zhang 2015 LIU, Y.-K.; HUANG, N.; ZHANG, Y.-M. Modeling of Human Welder Response Against 3D Weld Pool Surface Using Machine-Human Cooperative Virtualized Welding Platform. In: Tarn, TJ and Chen, SB and Chen, XQ (Ed.). *Robotic welding, intelligence and automation, RWIA'2014*. HEIDELBERGER PLATZ 3, D-14197 BERLIN, GERMANY: SPRINGER-VERLAG BERLIN, 2015. (Advances in Intelligent Systems and Computing, v. 363), p. 451–457. ISBN 978-3-319-18997-0; 978-3-319-18996-3. ISSN 2194-5357.

Lv et al. 2017 LV, N.; XU, Y.; LI, S.; YU, X.; CHEN, S. Automated control of welding penetration based on audio sensing technology. *Journal of Materials Processing Technology*, v. 250, p. 81–98, dec 2017. ISSN 09240136. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0924013617302777>.

Lv et al. 2016 LV, N.; XU, Y. L.; FANG, G.; YU, X. W.; CHEN, S. B. Research on welding penetration state recognition based on BP-Adaboost model for pulse GTAW welding dynamic process. In: *Proceedings of IEEE Workshop on Advanced Robotics and its Social Impacts, ARSO*. IEEE, 2016. v. 2016-Novem, p. 100–105. ISBN 9781509040773. ISSN 21627576. Available from Internet: http://ieeexplore.ieee.org/document/7736264.

Ma, Li and Chen 2017 MA, G.; LI, L.; CHEN, Y. Effects of beam configurations on wire melting and transfer behaviors in dual beam laser welding with filler wire. *Optics and Laser Technology*, Elsevier, v. 91, n. April 2016, p. 138–148, 2017. ISSN 00303992. Available from Internet: http://dx.doi.org/10.1016/j.optlastec.2016.12.019>.

Malviya and Pratihar 2011 MALVIYA, R.; PRATIHAR, D. K. Tuning of neural networks using particle swarm optimization to model MIG welding process. *Swarm and Evolutionary Computation*, Elsevier, v. 1, n. 4, p. 223–235, dec 2011. ISSN 2210-6502. Available from Internet: https://www.sciencedirect.com/science/article/abs/pii/S221065021100040X>.

Marbán, Mariscal and Segovia 2009 MARBÁN, Ó.; MARISCAL, G.; SEGOVIA, J. A Data Mining & Knowledge Discovery Process Model. *Data Mining and Knowledge*..., n. February, p. 1–17, 2009. ISSN 978-3-902613-53-0.

Marsland 2015 MARSLAND, S. *Machine Learning, An Algorithmic Perspective*. USA: CRC Press, 2015. ISBN 978-1-4665-8333-7.

Mathers 2002 MATHERS, G. Weld defects and quality control. In: MATHERS, G. (Ed.). *The Welding of Aluminium and its Alloys*. Woodhead Publishing, 2002, (Woodhead Publishing Series in Welding and Other Joining Technologies). p. 199–215. ISBN 978-1-85573-567-5. Available from Internet: https://www.sciencedirect.com/science/article/pii/B9781855735675500142>.

Mathew et al. 2017 MATHEW, J.; MOAT, R.; PADDEA, S.; FITZPATRICK, M.; BOUCHARD, P. Prediction of residual stresses in girth welded pipes using an artificial neural network approach. *International Journal of Pressure Vessels and Piping*, v. 150, p. 89–95, feb 2017. ISSN 03080161. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0308016115300971>.

McNamara, Green and Olsson 2006 MCNAMARA, J. M.; GREEN, R. F.; OLSSON, O. Bayes' theorem and its applications in animal behaviour. *Oikos*, Wiley/Blackwell (10.1111), v. 112, n. 2, p. 243–251, feb 2006. ISSN 00301299. Available from Internet: http://doi.wiley.com/10.1111/j.0030-1299.2006.14228. x>.

Mesnil et al. 2013 MESNIL, G.; HE, X.; DENG, L.; BENGIO, Y. Investigation of recurrent-neural-network architectures and learning methods for spoken language understanding. In: BIMBOT, F.; CERISARA, C.; FOUGERON, C.; GRAVIER, G.; LAMEL, L.; PELLEGRINO, F.; PERRIER, P. (Ed.). *Iterspeech*. [S.1.]: ISCA, 2013. p. 3771–3775.

Moncayo Torres 2013 Moncayo Torres, E. M. U. d. B. *Uma metodologia para modelagem e controle da largura do cordão de solda no processo gmaw*. Tese (Doutorado), 2013.

Mortimer 2013 MORTIMER, R. G. Chapter 15 - Probability, Statistics, and Experimental Errors. In: MORTIMER, R. G. (Ed.). *Mathematics for Physical Chemistry (Fourth Edition)*. Fourth edition. Boston: Elsevier, 2013. p. 191–206. ISBN 978-0-12-415809-2. Available from Internet: https://www.sciencedirect.com/science/article/pii/B978012415809200015X>.

Mota et al. 2013 MOTA, C. P.; MACHADO, M. V. R.; Finzi Neto, R. M.; VILARINHO, L. O. Sistema de visão por infravermelho próximo para monitoramento de processos de soldagem a arco. *Soldagem & Inspeção*, Associação Brasileira de Soldagem, v. 18, n. 1, p. 19–30, mar 2013. ISSN 0104-9224. Available from Internet: .

Muniategui et al. 2017 MUNIATEGUI, A.; HÉRIZ, B.; ECIOLAZA, L.; AYUSO, M.; ITURRIOZ, A.; QUINTANA, I.; ÁLVAREZ, P. Spot welding monitoring system based on fuzzy classification and deep learning. In: *2017 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE)*. IEEE, 2017. p. 1–6. ISBN 9781509060344. ISSN 10987584. Available from Internet: http://ieeexplore.ieee.org/document/8015618/>.

Murphy 2013 MURPHY, A. B. Influence of droplets in gas-metal arc welding: new modelling approach, and application to welding of aluminium. *Science and Technology of Welding and Joining*, v. 18, n. 1, p. 32–37, jan 2013. ISSN 1362-1718. Available from Internet: http://www.tandfonline.com/doi/full/10.1179/1362171812Y.0000000069>.

Muzaka et al. 2017 MUZAKA, K.; PARK, M. H.; LEE, J. P.; JIN, B. J.; LEE, B. R.; KIM, W. Y. I. S. A Study on Prediction of Welding Quality Using Mahalanobis Distance Method by Optimizing Welding Current for A Vertical-position Welding. In: *Procedia Engineering*. [s.n.], 2017. v. 174, p. 60–67. ISSN 18777058. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S1877705817301431.

Nandhitha 2016 NANDHITHA, N. M. Artificial Neural Network Based Prediction Techniques for Torch Current Deviation to Produce Defect-Free Welds in GTAW Using IR Thermography. In: . [S.l.: s.n.], 2016. p. 137–142.

NIST NIST. *Engineering Statistics Handbook*. Available from Internet: https://www.itl.nist.gov/div898/handbook/pri/section3/pri3361.htm.

Norton 1999 NORTON, M. J. Knowledge discovery in databases. *Library Trends*, v. 48, n. 1, p. 9–21, Summer 1999. Copyright - Copyright University of Illinois at Urbana, Graduate School of Library and Information Science, Publications Office Summer 1999; Última actualización - 2015-11-08; CODEN - LIBTA3. Available from Internet: https://search.proquest.com/docview/220463919?accountid=26646>.

Ogawa 2011 OGAWA, Y. High speed imaging technique Part 1-high speed imaging of arc welding phenomena. *Science and technology of welding and joining*, MANEY PUBLISHING, STE 1C, JOSEPHS WELL, HANOVER WALK, LEEDS LS3 1AB, W YORKS, ENGLAND, v. 16, n. 1, p. 33–43, 2011. ISSN 1362-1718.

Olson and Delen 2008 OLSON, D. L.; DELEN, D. *Advanced Data Mining Techniques*. 1st. ed. [S.l.]: Springer Publishing Company, Incorporated, 2008. ISBN 3540769161, 9783540769163.

Pachauri et al. 2014 PACHAURI, D.; KONDOR, R.; SARGUR, G.; SINGH, V. Permutation Diffusion Maps (PDM) with Application to the Image Association Problem in Computer Vision. 2014. 541–549 p.

Pachitariu et al. 2013 PACHITARIU, M.; PACKER, A. M.; PETTIT, N.; DALGLEISH, H.; HAUSSER, M.; SAHANI, M. *Extracting regions of interest from biological images with convolutional sparse block coding*. 2013. 1745–1753 p.

Pan and Yang 2010 PAN, S. J.; YANG, Q. A survey on transfer learning. *IEEE Trans. on Knowl. and Data Eng.*, IEEE Educational Activities Department, Piscataway, NJ, USA, v. 22, n. 10, p. 1345–1359, out. 2010. ISSN 1041-4347. Available from Internet: http://dx.doi.org/10.1109/TKDE.2009.191>.

Park and Kim 2017 PARK, J.; KIM, K.-Y. Prediction Modeling Framework With Bootstrap Aggregating for Noisy Resistance Spot Welding Data. *Journal of Manufacturing Science and Engineering*, v. 139, n. 10, p. 101003, aug 2017. ISSN 1087-1357. Available from Internet: http://manufacturingscience.asmedigitalcollection.asme.org/article.aspx?doi=10.1115/1.4036787>.

Pavan Kumar et al. 2017 Pavan Kumar, N.; DEVARAJAN, P. K.; Arungalai Vendan, S.; SHANMUGAM, N. Prediction of bead geometry in cold metal transfer welding using back propagation neural network. *The International Journal of Advanced Manufacturing Technology*, v. 93, n. 1-4, p. 385–392, oct 2017. ISSN 0268-3768. Available from Internet: http://link.springer.com/10.1007/s00170-016-9562-8>.

Pawlak 2002 PAWLAK, Z. Rough sets, decision algorithms and Bayes' theorem. *European Journal of Operational Research*, North-Holland, v. 136, n. 1, p. 181–189, jan 2002. ISSN 0377-2217. Available from Internet: https://www.sciencedirect.com/science/article/pii/S0377221701000297>.

Petković 2017 PETKOVIĆ, D. Prediction of laser welding quality by computational intelligence approaches. *Optik*, v. 140, p. 597–600, jul 2017. ISSN 00304026. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0030402617304928.

Piateski and Frawley 1991 PIATESKI, G.; FRAWLEY, W. *Knowledge Discovery in Databases*. Cambridge, MA, USA: MIT Press, 1991. ISBN 0262660709.

Piatetsky 2014 PIATETSKY, G. CRISP-DM, still the top methodology for analytics, data mining, or data science projects. 2014.

Pinto-Lopera et al. 2016 PINTO-LOPERA, J.; S. T. Motta, J.; Absi Alfaro, S.; PINTO-LOPERA, J. E.; S. T. Motta, J. M.; Absi Alfaro, S. C. Real-Time Measurement of Width and Height of Weld Beads in GMAW Processes. *Sensors*, Multidisciplinary Digital Publishing Institute, v. 16, n. 9, p. 1500, sep 2016. ISSN 1424-8220. Available from Internet: http://www.mdpi.com/1424-8220/16/9/1500>.

Qi and Guo 2014 QI, X.-G.; GUO, B. Determining common weights in data envelopment analysis with shannon's entropy. *Entropy*, v. 16, n. 12, p. 6394–6414, 2014. Copyright - Copyright MDPI AG 2014; Última actualización - 2015-01-20. Available from Internet: https://search.proquest.com/docview/1645147706?accountid=26646>.

Rajeev Kumar and Dr. Vandana Somkuva 2015 Rajeev Kumar and; Dr. Vandana Somkuva. A Review on Analysis, Monitoring and Detection of Weld Defect Products. *International Journal of Engineering Research and*, V4, n. 11, p. 664–667, 2015.

Ramanathan, Mangla and Satpathy 2018 RAMANATHAN, P.; MANGLA, K. K.; SATPATHY, S. Smart controller for conical tank system using reinforcement learning algorithm. *Measurement: Journal of the International Measurement Confederation*, v. 116, p. 422–428, feb 2018. ISSN 02632241. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0263224117307091.

Ramos, Carvalho and Absi Alfaro 2013 RAMOS, E. G.; CARVALHO, G. C. de; Absi Alfaro, S. C. Analysis of weld pool oscillation in P-GMAW by Means of Shadowgraphy Image Processing. *Soldagem & Inspeção*, ASSOC BRASIL SOLDAGEM, RUA DR GUILHERME BANNITZ NO 126 CONJ ITAIM-BIBI, SAO PAULO, SP 532-060, BRAZIL, v. 18, n. 1, p. 39–49, 2013. ISSN 0104-9224.

Rao, Srinivasa Rao and Deepak 2017 RAO, P. G.; Srinivasa Rao, P.; DEEPAK, B. B. GRNN-immune based strategy for estimating and optimizing the vibratory assisted welding parameters to produce quality welded joints. *Engineering Journal*, v. 21, n. 3, p. 251–267, jun 2017. ISSN 01258281. Available from Internet: http://www.engj.org/index.php/ej/article/view/1311/560>.

Redmon et al. 2016 REDMON, J.; DIVVALA, S.; GIRSHICK, R.; FARHADI, A. You Only Look Once: Unified, Real-Time Object Detection. *Cvpr 2016*, p. 779–788, 2016. ISSN 10636919.

Rhea et al. 2011 RHEA, C. K.; SILVER, T. A.; HONG, S. L.; RYU, J. H.; STUDENKA, B. E.; HUGHES, C. M. L.; HADDAD, J. M. Noise and complexity in human postural control: Interpreting the different estimations of entropy. *PLoS One*, v. 6, n. 3, 03 2011. Copyright - © 2011 Rhea et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited: Rhea CK, Silver TA, Hong SL, Ryu JH, Studenka BE, et al. (2011) Noise and Complexity in Human Postural Control: Interpreting the Different Estimations of Entropy. PLoS ONE 6(3): e17696. doi:10.1371/journal.pone.0017696; Última actualización - 2017-03-27. Available from Internet: .">https://search.proquest.com/docview/1292719130?accountid=26646>.

Rios-Cabrera et al. 2016 RIOS-CABRERA, R.; MORALES-DIAZ, A. B.; AVILES-VIÑAS, J. F.; LOPEZ-JUAREZ, I. Robotic GMAW online learning: issues and experiments. *International Journal of Advanced Manufacturing Technology*, v. 87, n. 5-8, p. 2113–2134, nov 2016. ISSN 14333015. Available from Internet: http://link.springer.com/10.1007/s00170-016-8618-0>.

Rong et al. 2016 RONG, Y.; HUANG, Y.; ZHANG, G.; CHANG, Y.; SHAO, X. Prediction of angular distortion in no gap butt joint using BPNN and inherent strain considering the actual bead geometry. *International Journal of Advanced Manufacturing Technology*, v. 86, n. 1-4, p. 59–69, sep 2016. ISSN 14333015. Available from Internet: http://link.springer.com/10.1007/s00170-015-8102-2>.

Rudner and Liang 2002 RUDNER, L. M.; LIANG, T. *Automated Essay Scoring Using Bayes' Theorem.* Technology and Assessment Study Collaborative, Caroline A. and Peter S. Lynch School

of Education, Boston, College, 2002. v. 1. 1–21 p. ISSN 1540-2525. Available from Internet: http://napoleon.bc.edu/ojs/index.php/jtla/article/view/1668>.

Sahu, Dey and Branco 2003 SAHU, S. K.; DEY, D. K.; BRANCO, M. D. A new class of multivariate skew distributions with applications to bayesian regression models. *Canadian Journal of Statistics*, Wiley-Blackwell, v. 31, n. 2, p. 129–150, jun 2003. ISSN 03195724. Available from Internet: http://doi.wiley.com/10.2307/3316064>.

Saini 1998 SAINI, B. Y. D. An Investigation of Gas Metal Arc Welding Sound Signature for On-Line Quality Control. n. April, p. 172–179, 1998.

Santhana Babu et al. 2016 Santhana Babu, A. V.; GIRIDHARAN, P. K.; Ramesh Narayanan, P.; Narayana Murty, S. V. S. Prediction of Bead Geometry for Flux Bounded TIG Welding of AA 2219-T87 Aluminum Alloy. *Journal of Advanced Manufacturing Systems*, v. 15, n. 02, p. 69–84, jun 2016. ISSN 0219-6867. Available from Internet: http://www.worldscientific.com/doi/abs/10.1142/S0219686716500074>.

Santos 2011 SANTOS, M. Un enfoque aplicado del control inteligente. *RIAI - Revista Iberoamericana de Automatica e Informatica Industrial*, v. 8, n. 4, p. 283–296, oct 2011. ISSN 16977920. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S1697791211000501>.

Sarkar et al. 2016 SARKAR, A.; DEY, P.; RAI, R.; SAHA, S. A comparative study of multiple regression analysis and back propagation neural network approaches on plain carbon steel in submerged-arc welding. *Sadhana - Academy Proceedings in Engineering Sciences*, Springer India, v. 41, n. 5, p. 549–559, 2016. ISSN 0973-7677. Available from Internet: .

Scotti, A; Ponomarev, V 2008 Scotti, A; Ponomarev, V. Soldagem MIG/MAG: melhor entendimento, melhor desempenho. São Paulo: Artliber, 2008. 284 p.

Sen 2015 SEN, M. Evaluation of Correlations between DP-GMAW Process Parameters and Bead Geometry. n. July, p. 265–279, 2015.

Seyyed and Mohammad 2011 SEYYED, A. H.; MOHAMMAD, B. N. Emotion recognition method using entropy analysis of eeg signals. *International Journal of Image, Graphics and Signal Processing*, v. 3, n. 5, p. 30–36, 08 2011. Copyright - Copyright Modern Education and Computer Science Press Aug 2011; Última actualización - 2014-11-21. Available from Internet: https://search.proquest.com/docview/1626715712?accountid=26646>.

Seyyedian Choobi, Haghpanahi and Sedighi 2012 Seyyedian Choobi, M.; HAGHPANAHI, M.; SEDIGHI, M. Prediction of welding-induced angular distortions in thin butt-welded plates using artificial neural networks. *Computational Materials Science*, Elsevier, v. 62, p. 152–159, sep 2012. ISSN 0927-0256. Available from Internet: https://www.sciencedirect.com/science/article/pii/S0927025612003035.

Shalev-Shwartz and Ben-David 2014 SHALEV-SHWARTZ, S.; BEN-DAVID, S. *Understanding Machine Learning: From Theory to Algorithms*. New York, USA: Cambridge University Press, 2014. ISBN 978-1-107-05713-5.

Sharma, Maheshwari and Rathee 2016 SHARMA, S. K.; MAHESHWARI, S.; RATHEE, S. Multi-Objective Optimization of Bead Geometry for Submerged Arc Welding of Pipeline Steel Using RSM-Fuzzy Approach. *Journal for Manufacturing Science and Production*, v. 16, n. 3, jan 2016. ISSN 2191-4184. Available from Internet: https://www.degruyter.com/view/j/ijmsp.2016.16.issue-3/jmsp-2016-0009/sml.

Shaw, Sicree and Zimmet 2010 SHAW, J. E.; SICREE, R. A.; ZIMMET, P. Z. Global estimates of the prevalence of diabetes for 2010 and 2030. *Diabetes research and clinical practice*, Elsevier, v. 87, n. 1, p. 4–14, 2010.

Siewert et al. 2014 SIEWERT, E.; WILHELM, G.; HAESSLER, M.; SCHEIN, J.; HANSON, T.; SCHNICK, M.; FÜSSEL, U. Visualization of Gas Flows in Welding Arcs by the Schlieren Measuring Technique. *Welding journal*, v. 93, n. January, p. 1–5, 2014.

Sik-Ho Tsang 2018 Sik-Ho Tsang. *Review: ResNet — Winner of ILSVRC 2015* (*Image Classification, Localization, Detection*) | by Sik-Ho Tsang | Towards Data Science. 2018. Available from Internet: https://towardsdatascience.com/ review-resnet-winner-of-ilsvrc-2015-image-classification-localization-detection-e39402bfa5d8>.

Smola, Smola and Schölkopf 2004 SMOLA, A. J.; SMOLA, A. J.; SCHÖLKOPF, B. A tutorial on support vector regression. 2004. Available from Internet: http://citeseerx.ist.psu.edu/viewdoc/summary?doi=10.1.1.114.4288>.

Sreedhar et al. 2012 SREEDHAR, U.; KRISHNAMURTHY, C. V.; BALASUBRAMANIAM, K.; RAGHUPATHY, V. D.; RAVISANKAR, S. Automatic defect identification using thermal image analysis for online weld quality monitoring. *Journal of materials processing technology*, ELSEVIER SCIENCE SA, PO BOX 564, 1001 LAUSANNE, SWITZERLAND, v. 212, n. 7, p. 1557–1566, jul 2012. ISSN 0924-0136.

Stanton 2012 STANTON, J. *An Introduction to Data Science*. [s.n.], 2012. 1–157 p. ISBN 9781491901427. Available from Internet: https://us.sagepub.com/en-us/nam/an-introduction-to-data-science/book256486>.

Sumesh et al. 2015 SUMESH, A.; RAMESHKUMAR, K.; MOHANDAS, K.; BABU, R. S. Use of machine learning algorithms for weld quality monitoring using acoustic signature. In: *Procedia Computer Science*. [s.n.], 2015. v. 50, p. 316–322. ISBN 9142226562. ISSN 18770509. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S1877050915005438>.

Sutton and Barto 2017 SUTTON, R.; BARTO, A. Reinforcement Learning: An Introduction. *Trends in Cognitive Sciences*, v. 3, n. 9, p. 360, 2017. ISSN 13646613. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S1364661399013315.

Szegedy et al. 2016 SZEGEDY, C.; IOFFE, S.; VANHOUCKE, V.; ALEMI, A. Inception-v4, Inception-ResNet and the Impact of Residual Connections on Learning. feb 2016. ISSN 01678655. Available from Internet: http://arxiv.org/abs/1602.07261.

Tamura et al. 2011 TAMURA, K.; PETERSON, D.; PETERSON, N.; STECHER, G.; NEI, M.; KUMAR, S. Mega5: molecular evolutionary genetics analysis using maximum likelihood, evolutionary distance, and maximum parsimony methods. *Molecular biology and evolution*, Oxford University Press, v. 28, n. 10, p. 2731–2739, 2011.

The Welding Master 2017 The Welding Master. *What is MIG Welding Process or GMAW* (*Gas Metal Arc Welding*)? 2017. Available from Internet: https://www.theweldingmaster.com/what-is-mig-welding-process/.

Thompson Martínez et al. 2021 Thompson Martínez, R.; Alvarez Bestard, G.; Martins Almeida Silva, A.; Absi Alfaro, S. C. Analysis of GMAW process with deep learning and machine learning techniques. *Journal of Manufacturing Processes*, Elsevier, v. 62, p. 695–703, feb 2021. ISSN 1526-6125.

Thomsen 2005 THOMSEN, J. Feedback linearization based arc length control for gas metal arc welding. In: *Proceedings of the 2005, American Control Conference, 2005.* IEEE, 2005. p. 3568–3573. ISBN 0-7803-9098-9. Available from Internet: http://ieeexplore.ieee.org/document/1470527/.

Tibshirani 1996 TIBSHIRANI, R. *Regression Shrinkage and Selection via the Lasso*. WileyRoyal Statistical Society, 1996. 267–288 p. Available from Internet: https://www.jstor.org/stable/2346178>.

Tibshirani 2013 TIBSHIRANI, R. Modern regression 1 : Ridge regression. In: . [S.l.: s.n.], 2013. p. 1–21.

Tuominen 2016 TUOMINEN, V. The measurement-aided welding cell—giving sight to the blind. *The International Journal of Advanced Manufacturing Technology*, v. 86, n. 1-4, p. 371–386, sep 2016. ISSN 0268-3768. Available from Internet: http://link.springer.com/10.1007/s00170-015-8193-9>.

United States Air Force commissioned United States Air Force commissioned. $IDEF \emptyset - Function$ Modeling Method – IDEF. Available from Internet: .

Varadan and Anastassiou 2006 VARADAN, V.; ANASTASSIOU, D. Inference of disease-related molecular logic from systems-based microarray analysis. *PLoS Computational Biology*, v. 2, n. 6, p. e68, 06 2006. Copyright - © 2006 Varadan and Anastassiou. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited: Varadan V, Anastassiou D (2006) Inference of Disease-Related Molecular Logic from Systems-Based Microarray Analysis. PLoS Comput Biol 2(6): e68. doi:10.1371/journal.pcbi.0020068; Última actualización - 2014-04-29. Available from Internet: ">https://search.proquest.com/docview/1314503028?accountid=26646>.

Version 2004 VERSION, D. Aalborg Universitet Advanced Control Methods for Optimization of Arc Welding Thomsen, J. S. 2004.

Villani, Modenesi and Bracarense 2016 VILLANI, P.; MODENESI, P. J.; BRACARENSE, A. Q. *Soldagem: Fundamentos e Tecnologia.* [S.I.]: Elsevier Brasil, 2016.

Wan et al. 2017 WAN, X.; WANG, Y.; ZHAO, D.; HUANG, Y. A comparison of two types of neural network for weld quality prediction in small scale resistance spot welding. *Mechanical Systems and Signal Processing*, v. 93, p. 634–644, sep 2017. ISSN 08883270. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0888327017300377>.

Wan et al. 2017 WAN, X.; WANG, Y.; ZHAO, D.; HUANG, Y. A.; YIN, Z. Weld quality monitoring research in small scale resistance spot welding by dynamic resistance and neural network. *Measurement: Journal of the International Measurement Confederation*, v. 99, p. 120–127, mar 2017. ISSN 02632241. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0263224116307114>.

Wang, Huang and Zhang 2004 WANG, G.; HUANG, P. G.; ZHANG, Y. M. Numerical analysis of metal transfer in gas metal arc welding under modified pulsed current conditions. *Metallurgical and Materials Transactions B*, v. 35, n. 5, p. 857–866, oct 2004. ISSN 1073-5615. Available from Internet: ">http://link.springer.com/10.1007/s1007/s1080/

Wang, Lü and Jing 2016 WANG, Y.; LÜ, X.; JING, H. Dynamic simulation of short-circuiting transfer in GMAW based on the "mass-spring" model. *International Journal of Advanced Manufacturing Technology*, v. 87, n. 1-4, p. 897–907, oct 2016. ISSN 14333015. Available from Internet: http://link.springer.com/10.1007/s00170-016-8538-z.

Wang 2014 WANG, Z. Monitoring of GMAW Weld Pool From the Reflected Laser Lines for Real-Time Control. *IEEE Transactions on industrial informatics*, IEEE-INST ELECTRICAL ELECTRONICS ENGINEERS INC, 445 HOES LANE, PISCATAWAY, NJ 08855-4141 USA, v. 10, n. 4, p. 2073–2083, nov 2014. ISSN 1551-3203.

Weaver 1949 WEAVER, S. *The mathematical theory of communication*. University of Illinois Press, 1949. 125 p. ISBN 0252725484. Available from Internet: https://dl.acm.org/citation.cfm?id=1102016>.

Witten and Frank 2005 WITTEN, I. H.; FRANK, E. *Data mining : practical machine learning tools and techniques*. Morgan Kaufman, 2005. 525 p. ISBN 008047702X. Available from Internet: https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions-.

Witten and Frank 2005 WITTEN, I. H. I. H.; FRANK, E. *Data mining : practical machine learning tools and techniques*. Morgan Kaufman, 2005. 525 p. ISBN 008047702X. Available from Internet: ">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es&source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es@source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es@source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es@source=gbs_book_other_versions>">https://books.google.co.id/books?id=QTnOcZJzlUoC&hl=es@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gbs_book_other_versions>">https://books@source=gb

Wollstadt et al. 2014 WOLLSTADT, P.; MARTINEZZARZUELA, M.; VICENTE, R.; DIAZ-PERNAS FRANCISCO, J.; WIBRAL, M. Efficient transfer entropy analysis of non-stationary neural time series. *PLoS One*, v. 9, n. 7, p. e102833, 07 2014. Copyright - © 2014 Wollstadt et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited: Wollstadt P, Martínez-Zarzuela M, Vicente R, Díaz-Pernas FJ, Wibral M (2014) Efficient Transfer Entropy Analysis of Non-Stationary Neural Time Series. PLoS ONE 9(7): e102833. doi:10.1371/journal.pone.0102833; Última actualización - 2014-09-13. Available from Internet: ">https://search.proquest.com/docview/1548811346?accountid=26646>.

Woods and La 2019 Woods, A. C.; La, H. M. A novel potential field controller for use on aerial robots. *IEEE Transactions on Systems, Man, and Cybernetics: Systems*, v. 49, n. 4, p. 665–676, April 2019. ISSN 2168-2216.

Wu et al. 2016 WU, D.; CHEN, H.; HE, Y.; SONG, S.; LIN, T.; CHEN, S. A prediction model for keyhole geometry and acoustic signatures during variable polarity plasma arc welding based on extreme learning machine. *Sensor review*, EMERALD GROUP PUBLISHING LTD, HOWARD HOUSE, WAGON LANE, BINGLEY BD16 1WA, W YORKSHIRE, ENGLAND, v. 36, n. 3, p. 257–266, 2016. ISSN 0260-2288.

Wu et al. 2017 WU, D.; HUANG, Y.; CHEN, H.; HE, Y.; CHEN, S. VPPAW penetration monitoring based on fusion of visual and acoustic signals using t-SNE and DBN model. *Materials and Design*, v. 123, p. 1–14, jun 2017. ISSN 18734197. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0264127517302721.

Wu et al. 2014 WU, Q.-Q.; LEE, J.-P.; PARK, M.-H.; PARK, C.-K.; KIM, I.-S. A study on Development of Optimal Noise Filter Algorithm for Laser Vision System in GMA Welding. In: Xavior, MA and Yarlagadda, PKDV (Ed.). *12th global congress on manufacturing and management (GCMM - 2014)*. SARA BURGERHARTSTRAAT 25, PO BOX 211, 1000 AE AMSTERDAM, NETHERLANDS: ELSEVIER SCIENCE BV, 2014. (Procedia Engineering, v. 97), p. 819–827. ISSN 1877-7058.

Xu et al. 2014 XU, Y.; FANG, G.; CHEN, S.; ZOU, J. J.; YE, Z. Real-time image processing for vision-based weld seam tracking in robotic GMAW. *International journal of advanced manufacturing technology*, SPRINGER LONDON LTD, 236 GRAYS INN RD, 6TH FLOOR, LONDON WC1X 8HL, ENGLAND, v. 73, n. 9-12, p. 1413–1425, 2014. ISSN 0268-3768.

Xu et al. 2012 XU, Y.; YU, H.; ZHONG, J.; LIN, T.; CHEN, S. Real-time image capturing and processing of seam and pool during robotic welding process. *Industrial robot-an international journal*,

EMERALD GROUP PUBLISHING LIMITED, HOWARD HOUSE, WAGON LANE, BINGLEY BD16 1WA, W YORKSHIRE, ENGLAND, v. 39, n. 5, p. 513–523, 2012. ISSN 0143-991X.

Yan 2011 YAN, Z. Simulation and controlling for weld shape process in P-GMAW based on fuzzy logic. In: 2011 IEEE International Conference on Mechatronics and Automation, ICMA 2011. IEEE, 2011. p. 2078–2082. ISBN 9781424481149. ISSN 2152-7431. Available from Internet: http://ieeexplore.ieee.org/document/5986301/.

Yin, Yu and Zhou 2018 YIN, L.; YU, T.; ZHOU, L. Design of a Novel Smart Generation Controller Based on Deep Q Learning for Large-Scale Interconnected Power System. *Journal of Energy Engineering*, v. 144, n. 3, p. 04018033, jun 2018. ISSN 07339402. Available from Internet: ">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}29EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}39EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}39EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}28ASCE{\%}39EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}39EY.1943-7897.0>">http://ascelibrary.org/doi/10.1061/{\%}30

You, Gao and Katayama 2015 YOU, D.; GAO, X.; KATAYAMA, S. WPD-PCA-Based Laser Welding Process Monitoring and Defects Diagnosis by Using FNN and SVM. *IEEE Transactions on industrial electronics*, IEEE-INST ELECTRICAL ELECTRONICS ENGINEERS INC, 445 HOES LANE, PISCATAWAY, NJ 08855-4141 USA, v. 62, n. 1, p. 628–636, 2015. ISSN 0278-0046.

Yu and Deng 2011 YU, D.; DENG, L. Deep Learning and Its Applications to Signal and Information Processing [Exploratory DSP. *IEEE Signal Processing Magazine*, v. 28, n. 1, p. 145–154, jan 2011. ISSN 1053-5888. Available from Internet: http://ieeexplore.ieee.org/document/5670617/.

Yue et al. 2009 YUE, H.; LI, K.; ZHAO, H. W.; ZHANG, Y. Vision-based pipeline girth-welding robot and image processing of weld seam. *Industrial Robot-an International Journal*, v. 36, n. 3, p. 284–289, may 2009. ISSN 0143991X. Available from Internet: http://www.emeraldinsight.com/doi/10.1108/01439910910950568>.

Zellner and Chetty 1965 ZELLNER, A.; CHETTY, V. K. Prediction and Decision Problems in Regression Models from the Bayesian Point of View. *Journal of the American Statistical Association*, v. 60, n. 310, p. 608–616, jun 1965. ISSN 0162-1459. Available from Internet: http://www.tandfonline.com/doi/abs/10.1080/01621459.1965.10480817>.

Zhang YuMing 2008 ZHANG YUMING, M. *Real-time weld process monitoring*. Woodhead Pub. and Maney Pub. on behalf of the Institute of Materials, Minerals and Mining, 2008. 299 p. ISBN 9781845692681. Available from Internet: http://www.sciencedirect.com/science/book/9781845692681>

Zhao and Chung 2018 ZHAO, Y.; CHUNG, H. Numerical simulation of the transition of metal transfer from globular to spray mode in gas metal arc welding using phase field method. *Journal of Materials Processing Technology*, v. 251, p. 251–261, jan 2018. ISSN 09240136. Available from Internet: http://linkinghub.elsevier.com/retrieve/pii/S0924013617303953>.

Zhou 2003 ZHOU, Z.-H. Three perspectives of data mining. *Artificial Intelligence*, v. 143, n. 1, p. 139–146, 2003. ISSN 00043702. Available from Internet: http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.2.3790 (& rep=rep1 & ty>.

Zhu et al. 2014 ZHU, Z.; LUO, P.; WANG, X.; TANG, X. *Multi-View Perceptron: a Deep Model for Learning Face Identity and View Representations*. 2014. 217–225 p.

ANNEXES

A. WELD-BEAD GEOMETRY CALC CODE

```
1 import imageio
2 from matplotlib import pyplot as plt
3
4 # initial data depth high
5 #path_img = test2_pronta.png
6 \ \#h\_img = 310
7 #coord_left = (145, 155)
8 #coord_right = (3549, 74)
9 #pieceh_left_px = 125
10 #pieceh_right_px = 127
11 #pieceh_left_mm = 7
12 #pieceh_right_mm = 7
13 #width = False
14 #name_file = file_depth_high.csv
15 #color_high = [10, 54, 128]
16 #color_depth = [150, 6, 6]
17
18 # initial data width
19 path_img = test3_pronta.png
20 h_{img} = 432
21 \text{ coord\_left} = (90, 365)
22 coord_right = (3912, 268)
23 pieceh_left_px = 516
24 pieceh_right_px = 530
25 pieceh_left_mm = 23
26 width = True
27 name_file = file_width.csv
28 \text{ color}_high = [150, 6, 6]
29
30 # initial data
31 img_solda = imageio.imread(path_img)
32
33
34 # all functions
35
36
  def calc_funcline():
37
       if coord_right[0] != coord_left[0]:
38
           m = (h_{img} - coord_{left[1]} - (h_{img} - coord_{right[1]})) / 
39
                      (coord_left[0] - coord_right[0])
40
41
           n = coord_left[1] - m * coord_left[0]
       else:
42
           print( error )
43
44
       return m, n
45
46
47 def values_line(m, n):
```

```
values = []
48
       for x in range(coord_left[0], coord_right[0]):
49
           # images coord are inverse
50
51
           values.append(int(n - m * x ))
       return values
52
53
54
55
  def values_high(line):
       values = []
56
       values.append(0)
57
       for i, pos in enumerate(line):
58
59
           y_pos = pos
           x_pos = coord_left[0] + i
60
           while y_pos > 0 and not (img_solda[y_pos, x_pos] == color_high).all():
61
62
                y_pos -= 1
           if (img_solda[y_pos, x_pos] == color_high).all():
63
                img_solda[y_pos - 1, x_pos] = color_high
64
                img_solda[y_pos - 2, x_pos] = color_high
65
               img_solda[y_pos - 3, x_pos] = color_high
66
               values.append(pos - y_pos)
67
           elif y_pos == 0:
68
               values.append(0)
69
70
       values.append(0)
71
       plt.imshow(img_solda)
72
       plt.show()
73
       return values
74
75
76
  def values_depth(line):
77
       values = []
78
       values.append(0)
79
       for i, pos in enumerate(line):
80
81
           y_pos = pos
           x_pos = coord_left[0] + i
82
           while y_pos < h_img and not (img_solda[y_pos, x_pos] == color_depth).all():</pre>
83
84
               y_pos += 1
                # print(y_pos)
85
           if (img_solda[y_pos, x_pos] == color_depth).all():
86
                img_solda[y_pos + 1, x_pos] = color_depth
87
88
                img_solda[y_pos + 2, x_pos] = color_depth
               img_solda[y_pos + 3, x_pos] = color_depth
89
                values.append(pos - y_pos)
90
           elif y_pos == h_img:
91
               values.append(0)
92
           # print(pos, y_pos)
93
94
       values.append(0)
       # plt.imshow(img_solda)
95
       # plt.show()
96
97
       return values
98
99
```

```
def values_geo(line):
100
101
        depth = []
        if not width:
102
103
            depth = values_depth(line)
       high = values_high(line)
104
       return depth, high
105
106
107
   def calc_coef(index):
108
109
        # pixcel ajust
        coef = pieceh_left_px * (pieceh_right_px - pieceh_left_px) / (float(
110
            (coord_right[0] - coord_left[0]) * index + pieceh_left_px *
111
112
            (pieceh_right_px - pieceh_left_px)))
113
        return coef
114
115 def calc_func_deformation():
        if coord_right[0] != coord_left[0]:
116
117
            m = (pieceh_right_px - pieceh_left_px) / ( coord_right[0] - coord_left[0])
118
       else:
119
            print( error )
120
        return m
121
122 def calc_pieceh_px(m, n, index):
123
        # pixcel ajust
       pieceh_px = m \star index + n
124
        return pieceh_px
125
126
127 def calc_mm_ajust(value):
        return value * pieceh_left_mm / pieceh_left_px
128
129
130
   def calc_mm(value, pieceh_px):
        return value * pieceh_left_mm / pieceh_px
131
132
   def save_csv(depth, high):
133
       m = calc_func_deformation()
134
135
       n = pieceh_left_px
       with open(name_file, w ) as f:
136
            i = 0
137
            if not width:
138
                f.write(depth, high \n)
139
140
                for v_depth, v_high in zip(depth, high):
                     pieceh_px = calc_pieceh_px(m, n, i)
141
                     mm_depth = calc_mm(v_depth, pieceh_px)
142
                     mm_high = calc_mm(v_high, pieceh_px)
143
                     f.write(str(mm_depth) + , +
144
                              str(mm_high) + \langle n \rangle
145
146
                     i += 1
            else:
147
                f.write(width \n)
148
149
                for v_high in high:
                     pieceh_px = calc_pieceh_px(m, n, i)
150
151
                     mm_width = calc_mm(v_high, pieceh_px)
```

```
152
                    f.write(str(mm_width) + \n)
153
                    i += 1
154
155
           f.close()
156
157
158
  if __name__ == __main__ :
159
       m, n = calc_funcline()
160
       print(" calculated m and n")
161
162
       print(m, n)
       line = values_line(m, n)
163
       plt.plot(line, r , linewidth=1)
164
165
       plt.show()
       print(" calculated line")
166
       depth, high = values_geo(line)
167
       print(" calculated geometry")
168
169
       print(len(depth), len(high))
       save_csv(depth, high)
170
171
       print(" saved geometry")
```

B. CONVOLUTIONAL NEURAL NETWORK CODE

```
from fastai.imports import *
1
2
  from fastai.transforms import *
3 from fastai.conv_learner import *
4 from fastai.model import *
  from fastai.dataset import *
5
6 from fastai.sgdr import *
7 from fastai.plots import *
   from tqdm import tqdm
8
  tqdm.monitor_interval = 0
9
10 PATH = "../../metal_drop/data/"
   sz=96
11
   arch=resnet18
12
   bs=64
13
14
   augs = transforms_side_on + [AddPadding(pad=10, mode=cv2.BORDER_REPLICATE)]
15
   tfms = tfms_from_model(arch, sz, aug_tfms=augs, max_zoom=1.1)
16
   data = ImageClassifierData.from_paths(PATH, tfms=tfms, bs=bs)
17
   learn = ConvLearner.pretrained(arch, data, precompute=False, ps=0.4)
18
   learn.opt_fn=optim.Adam
19
20
   %time learn.fit(1e-2, 3, cycle_len=1, cycle_mult=2)
21
22
  learn.unfreeze()
23
24 lr = np.array([1e-3, 1e-3, 1e-2])
25
  %time learn.fit(lr, 3, cycle_len=1, cycle_mult=2)
```

C. UNMELTED WIRE CALC CODE

```
def calc_wire_width(image_array, x, y, xl_ant = 10, xr_ant=15):
1
       .....
2
       User give coord(x, y) in the wire for algorithm search in the same line
3
       wire init width and it value.
4
5
       image_array: image for wire calc
6
       x,y: coordenate between wire side
7
       ....
8
       left = -1
9
       right = 1
10
       value_left = x
11
       value_right = x
12
       flag_left = False
13
       flag_right = False
14
       resized_image = cv2.resize(np.uint8(image_array), (32, 32))
15
       thresh = edge_detect(resized_image)
16
17
       while not(flag_left and flag_right):
18
           if not flag_left:
19
               left -= 1
20
               if thresh[y, x + left] == 255:
21
                    value_left = x + left
22
                    flag_left = True
23
               elif (x + left) == (xl_ant-2):
24
                    value_left = xl_ant
25
                    flag_left = True
26
27
28
           if not flag_right:
29
               right += 1
30
               if thresh[y, x + right] == 255:
31
                    value_right = x + right
32
                    flag_right = True
33
               elif (x + right) == (xr_ant+2):
34
35
                    value_right = xr_ant
                    flag_right = True
36
       return value_right, value_left
37
38
    def calc_unmelted_wire(image_array, h, x1, xr, base_h, e=1, wire_d=1.2, bico=15):
39
       .....
40
41
       It detect and calc unmelted wire high dimention
42
       image_array: image for wire calc
43
       h: it is high init for search stick out (px)
44
       xl: x left of wire (px)
45
       xr: x right of wire (px)
46
47
       base_h: high of base system(px)
```

```
48
       e: allowed error in width dimention
49
       wire_d: wire diameter (mm)
       bico: dimention base to bico (mm)
50
       ....
51
52
       temp_xl = xl
53
       temp_xr = xr
54
55
       flag = True
56
57
       thresh = edge_detect_resize(np.uint8(image_array))
58
       while ((xr - xl) \leq (temp_xr - temp_xl) + e) and h \leq (base_h - 2) and flag:
59
           h += 1
60
           if thresh[h, temp_xl] != 0:
61
               if thresh[h, temp_xl + 1] == 0:
62
                    temp_xl += 1
63
64
65
           if thresh[h, temp_xr] != 0:
                if thresh[h, temp_xr - 1] == 0:
66
                    temp_xr -= 1
67
68
69
           # it detect if temp values worked
70
71
           if thresh[h, temp_xl] != 0 or thresh[h, temp_xr] != 0:
                flag = False
72
73
       h_px = base_h - h-1
74
       dx = xr - xl
75
       if dx == 0:
76
           dx = 1
77
       h_mm = h_px * wire_d / dx
78
       unmelted_wire = bico - h_mm
79
80
       return h - 5, unmelted_wire
81
```

D. PREDICTION CODE

```
1 import pandas as pd
2 from sklearn.cross_validation import train_test_split
3 from sklearn.metrics import mean_squared_error
4 # from fancyimpute import KNN
5 from knn_impute import knn_impute
6 from sklearn.externals import joblib
7 # algorithms
8 from sklearn.multioutput import MultiOutputRegressor
9 from sklearn.preprocessing import PolynomialFeatures
10 from sklearn.pipeline import make_pipeline
11 from sklearn.linear_model import Lasso
12 from sklearn.linear_model import ElasticNet
13 from sklearn.linear_model import LinearRegression
14 from sklearn.linear_model import Ridge
15 from sklearn.linear_model import LassoLars
16 from sklearn.linear_model import BayesianRidge
17 from sklearn.svm import SVR
18 from sklearn.neural_network import MLPRegressor
19
20 # load dataset
21 df_exp1 = pd.read_csv( parte 1/final/aws_a5_1_all.csv )
22 df_exp2 = pd.read_csv( parte 2/final/aws_a5_2_all.csv )
23 df_aws = pd.concat([df_exp1, df_exp2])
24 df_aws = df_aws.drop([ image , time , cputime_dif , num ], 1)
25
26
  # loss data
27
  if df_aws[ wire_l ].isna().any():
28
       dfwire_l = knn_impute(target=df_aws[ wire_l ],
29
                 attributes=df_aws.drop([ wire_l , volume ], 1),
30
31
                     aggregation_method="median",
                             k_neighbors=10, numeric_distance= euclidean ,
32
                             categorical_distance= hamming
33
                             missing_neighbors_threshold=0.8)
34
35
      df_aws[ wire_l ] = dfwire_l
36
      dfvolume = knn_impute(target=df_aws[ volume ],
37
                 attributes=df_aws.drop([ volume ], 1),
38
                             aggregation_method="median", k_neighbors=10,
39
                             numeric_distance= euclidean ,
40
41
                             categorical_distance= hamming ,
                             missing_neighbors_threshold=0.8)
42
      df_aws[ volume ] = dfvolume
43
44
45 # run model
46 X = df_aws.drop([ high , width , depth ], 1)
47 y = df_aws.drop([ tension , velocity , weld_velocity , wire_l , volume ], 1)
```

```
# estimators
48
  ESTIMATORS = {
49
       "Linear regression": LinearRegression(),
50
       "Ridge": Ridge(alpha=1.802e-01),
51
       "Lasso": Lasso(alpha=1.802e-01),
52
      "ElasticNet": ElasticNet(random_state=0, alpha=1.802e-01),
53
       "LARS Lasso": LassoLars(alpha=1.802e-01),
54
       "Bayesian Regression": BayesianRidge(),
55
      "SVR poly": SVR(kernel= poly , C=1e3),
56
       "SVR rbf": SVR(kernel= rbf, C=1e3),
57
58
       }
59
60 # Create an empty dictionary to collect prediction values
61 y_test_predict = dict()
62 \text{ y_mse} = \text{dict}()
63 \text{ mim}_y = 2.0
64 df_model_error = pd.DataFrame(index=[*ESTIMATORS])
65
  for i in range(1, 7):
66
       iteration = iter +str(i)
67
       X_train, X_test, y_train, y_test = train_test_split(X, y, test_size=0.33,
68
69
                                random_state=42)
       for name, estimator in ESTIMATORS.items():
70
       if name == SVR poly :
71
               multiestimator = MultiOutputRegressor(SVR(kernel= poly ,
72
                                  C=1e3, degree=i))
73
           elif name == SVR rbf :
74
              multiestimator = MultiOutputRegressor(estimator)
75
           else:
76
         multi = MultiOutputRegressor(estimator)
77
78
         multiestimator = make_pipeline(PolynomialFeatures(i), multi)
       # fit() with instantiated object
79
           multiestimator.fit(X_train, y_train)
80
81
           # Make predictions and save it in dict under key: name
82
           y_test_predict[name] = multiestimator.predict(X_test)
83
84
           y_mse[name] = float(mean_squared_error(y_test,
                               multiestimator.predict(X_test)))
85
           print( RMSE for , name, were , y_mse[name])
86
           y_round = round(y_mse[name],2)
87
88
           filename = models/ +name+ _ +str(iteration) + .model
           # save model
89
           joblib.dump(multiestimator, filename)
90
91
       df_model_error[str(i)] = pd.Series(y_mse, index=df_model_error.index)
92
93
94 df_model_error.to_csv( model_error.csv )
```

E. REINFORCEMENT LEARNING CODE

```
i import random
2 import numpy as np
3 import pandas as pd
4
5 class Qtable:
6
      num_episodes = 10000
7
8
      max_step_per_episode = 100
9
       learning_rate = 0.1
10
      discount_rate = 0.99
11
12
       exploration_rate = 1
13
      max_exploration_rate = 1
14
      min_exploration_rate = 0.01
15
       exploration_decay_rate = 0.001
16
17
       rewards_all_episodes = []
18
19
       df_exp1 = pd.read_csv( parte 1/final/aws_a5_1_all.csv )
20
       df_exp2 = pd.read_csv( parte 2/final/aws_a5_2_all.csv )
21
       df_aws = pd.concat([df_exp1, df_exp2])
22
23
      def __init__(self, state, init_action_values):
24
25
           self.__obj_state = state
26
           self.__values = init_action_values
27
           self.___init = init_action_values
28
           # values min and max tension , wire rate , welding velocity
29
           self.__extrem_val =[(17,32),(4.8,8.2),(6.2,13.4)]
30
31
           # making q-table
           df_state = self.df_aws[[ high , width , depth ]]
32
           # index states < high , width , depth >
33
           self.state_list ={}
34
35
           # columns actions < tension , wire rate , welding velocity >
36
           self.action_list = [ +++ , --- , === , ++= , =++ , +=+ , -++ , +-+ ,
37
                                 ++- , +-- , -+- , --+ , +== , =+= , ==+ , =-- ,
38
                                 ---- , ---= , -== , =-- , -+= , -=+ , +-= ,
39
                                 =-+ , =+- , +=- ]
40
41
           self.__q_table = pd.DataFrame(columns=self.action_list)
42
43
44
       def calc_reward(self, state):
           dif = 0
45
           # permited error
46
47
           e = 0
```

```
for i in range(0, len(self.__obj_state)):
48
               dif += abs(self.__obj_state[i]-state[i])
49
           return -1*(dif - e) + 10
50
51
       def get_action(self, state):
52
           # get new action
53
           return pd.Series.idxmax(self.__q_table.loc[state,:])
54
55
       def get_weld_values(self, action):
56
           # variation are values for eachs welding parameter variation
57
           \# variation = (3, 1, 2)
58
59
           variation = (2, 1, 1)
           result = [0, 0, 0]
60
           for index, act in enumerate(action):
61
               if act == + :
62
                    result[index] = self.__values[index] + variation[index]
63
               elif act == - :
64
                    result[index] = self.__values[index] - variation[index]
65
66
               if result[0] < self.__extrem_val[0][0]:</pre>
67
                    result[0] = self.__extrem_val[0][0]
68
               if result[0] > self.__extrem_val[0][1]:
69
                    result[0] = self.__extrem_val[0][1]
70
               if result[1] < self.__extrem_val[1][0]:</pre>
71
                    result[1] = self.__extrem_val[1][0]
72
               if result[1] > self.__extrem_val[1][1]:
73
                    result[1] = self.__extrem_val[1][1]
74
               if result[2] < self.__extrem_val[2][0]:</pre>
75
                    result[2] = self.__extrem_val[2][0]
76
               if result[2] > self.__extrem_val[2][1]:
77
                    result[2] = self.__extrem_val[2][1]
78
79
               self.___values = result
80
           return result
81
82
       def check_state_exist(self, state):
83
84
           string = str(state[0])+str(state[1])+str(state[2])
           if string not in self.state_list.keys():
85
                # append new state to q table
86
               self.state_list[string] = state
87
               self.__q_table = self.__q_table.append(
88
                    pd.Series(
89
                        [-10] *len(self.action_list),
90
                        index=self.__q_table.columns,
91
                        name=string,
92
                    )
93
94
               )
       # method for offline train
95
       def get_volume_wirel(self, tupla):
96
           df_temp = self.df_aws.loc[(self.df_aws[ tension ] -
97
                        tupla[0]).abs().argsort()].dropna()
98
           df_temp = df_temp.loc[(df_temp[ velocity ] -
99
```

```
100
                       tupla[1]).abs().argsort()].dropna()
101
            df_temp = df_temp.loc[(df_temp[ weld_velocity ] -
                       tupla[2]).abs().argsort(),
102
103
                                          [ volume , wire_l ]].dropna()
            return df_temp.iloc[0][ volume ], df_temp.iloc[0][ wire_1 ]
104
105
       def prox_value(self, tuple):
106
107
            dict_list ={}
            for index, row in enumerate(self.state_list):
108
109
                dict_list[round(abs(row[0] - tuple[0]) + abs(row[1] - tuple[1]) + \
                  abs(row[2] - tuple[2]), 2)] = index
110
            sort_dic =sorted(dict_list)
111
112
            return dict_list[sort_dic[0]]
113
114
       def find_dict_key(self, state):
            for key, val in self.state_list.items():
115
                if state == val:
116
117
                    return key
118
       def learn(self, name, env):
119
                Q learning algorithm
120
            # states ( high , width , depth )
121
            state = (2.18, 5.48, -2.95)
122
            self.check_state_exist(state)
123
            for episode in range(self.num_episodes):
124
125
126
                reward_current_episode = 0
127
                for step in range(self.max_step_per_episode):
128
                    # exploration-explotation trade-off
129
                    exploration_rate_threshold = random.uniform(0, 1)
130
                    key_state = self.find_dict_key(state)
131
132
                    if exploration_rate_threshold > self.exploration_rate:
                         index = np.argmax(self.__q_table.loc[key_state,:])
133
                        action = self.__q_table.columns[index]
134
                    else:
135
136
                         index = np.random.choice(len(self.action_list))
                         action = self.action_list[index]
137
138
                    weld_parameters = self.get_weld_values(action)
139
140
                    # learning offline
                    vol, wire = self.get_volume_wirel(weld_parameters)
141
                    # tension velocity
                                            volume weld_velocity wire_l
142
                    weld_parameters.insert(2,vol)
143
                    weld_parameters.append(wire)
144
145
146
                    # return < depth , high , width >
                    new_state = env.predict([weld_parameters])
147
                    new_state = [round(new_state[0][1], 2),
148
149
                     round(new_state[0][2], 2),
                     round(new_state[0][0], 2)]
150
151
                    reward = round(self.calc_reward(state),2)
```

```
152
                     # Update Q-table for Q(s, a)
153
                     print(
                             step %s of %s ,(step, self.max_step_per_episode))
154
                     self.check_state_exist(new_state)
                     key_state_new = self.find_dict_key(new_state)
155
                     self.__q_table.loc[key_state, action] = self.__q_table.loc[key_state,
156
                                              action] * \setminus
157
                                    (1 - self.learning_rate) + \setminus
158
159
                                     self.learning_rate * (reward +
                                     self.discount_rate *
160
                                     np.max(self.__q_table.loc[
161
162
                                                 key_state_new, :]))
163
                     state = new_state
164
165
166
                     reward_current_episode += reward
167
                     if reward >= 10:
168
169
                         break
170
171
                # Exploration rate decay
                self.exploration_rate = self.min_exploration_rate + \
172
                     (self.max_exploration_rate - self.min_exploration_rate) * \
173
174
                     np.exp(-self.exploration_decay_rate * episode)
175
                self.rewards_all_episodes.append(reward_current_episode)
176
```